

# Success in the New Welfare Environment: An Assessment of Approaches in HUD's Employment and Training Initiatives

March 2000

# **Success in the New Welfare Environment: An Assessment of Approaches in HUD's Employment and Training Initiatives**

Prepared for:

U.S. Department of Housing and Urban Development Office of Policy Development and Research

Authors:

Michael E. Fishman Kathleen A. Boland Karen N. Gardiner Abby Sosland

Prepared by:

ICF Consulting The Lewin Group, Inc.

Contract DU100C000005966

March 2000

# Success in the New Welfare Environment: An Assessment of Approaches in HUD's Employment and Training Initiatives

Prepared for:

U.S. Department of Housing and Urban Development Office of Policy Development and Research

Prepared by:

ICF Consulting Kathleen A. Boland Abby Sosland The Lewin Group, Inc. Michael E. Fishman Karen N. Gardiner

Contract DU100C000005966

March 2000

INTRODUCTIONI-1 CHAPTER I: EMPLOYMENT IS THE HALLMARK OF THE NEW WELFARE ENVIRONMENT
A. PRWORA FUNDAMENTALLY CHANGED WELFARE RULES, CREATING A STRONG "WORK FIRST" ENVIRONMENT
ENVIRONMENT
B. STATE CHARACTERISTICS AND POLICIES
C. IMPACT OF WELFARE REFORM ON PUBLIC HOUSING RESIDENTS OR FAMILIES RECEIVING SECTION 8 ASSISTANCE
D. IMPACT ON PUBLIC HOUSING AUTHORITIES
E. IMPACT ON HOUSING COMMUNITIES
CHAPTER II: THE IMPORTANT ROLE OF HUD IN SUPPORTING TRANSITION TO WORK
A. PROGRAM HISTORY
B. NEW LEGISLATIVE MANDATES AND INITIATIVES REFLECT CHANGING ENVIRONMENT
C. CREATING SUCCESSFUL HOUSING AND WELFARE STRATEGIES AT THE LOCAL LEVEL
CHAPTER III: HOUSING AUTHORITY EMPLOYMENT AND TRAINING SERVICES FOR TANF RESIDENTS OFTEN INVOLVE WELFARE AND LABOR AGENCIES
A. INTRODUCTION
A. INTRODUCTION
C. HUD PROGRAMS GENERALLY DO NOT COUNT TOWARD TANF WORK REQUIREMENTS
D. PRIMARY FOCUS OF COLLABORATION ON EMPLOYMENT AND TRAINING SERVICES FOR TANF RECIPIENTS IS
OFTEN OUTSIDE OF DISCRETE HUD PROGRAMS
E. JOBS PLUS PROVIDES A STRONG FOCUS ON COLLABORATION WITH WELFARE AND LABOR AGENCIES
F. LESS EVIDENCE OF COORDINATION AND COLLABORATION FOR PROGRAMS THAT TARGET SECTION 8 RESIDENTS
CHAPTER IV: HUD PROGRAM ALIGNMENT WITH FINDINGS FROM EVALUATIONS OF EMPLOYMENT AND TRAINING PROGRAMS
A. INTRODUCTION
B. PROGRAM PHILOSOPHY
C. JOB SEARCH
E. VOCATIONAL TRAINING
F. BASIC SKILLS TRAINING
G. POST SECONDARY EDUCATION
H. ASSESSMENT
I. CASE MANAGEMENT
J. BARRIERS TO EMPLOYMENT
K. CASH AND IN-KIND SUPPORTS
CHAPTER V: SUMMARY OF FINDINGS AND CONCLUSIONS
A. SUMMARY OF FINDINGS
B. CONCLUSIONS
BIBLIOGRAPHY
APPENDIX A: ASSESSMENT PLAN

# **Table of Contents**

# TABLES AND FIGURES

HUD PROGRAM COMPONENTS SITE CHARACTERISTICS CHART TABLE 1.1: STATE TANF PROGRAM OVERVIEW: TABLE 1.2: NUMBER OF INDIVIDUALS ON AFDC/TANF BY STATE TABLE 4.1: ASSESSMENT FRAMEWORK FIGURE 1: SYSTEMS APPROACH EXECUTIVE SUMMARY, PAGE 5 INTRODUCTION, PAGE 4 CHAPTER 1, PAGE 3 CHAPTER 1, PAGE 5 CHAPTER 4, PAGE 31 CHAPTER 4, PAGE 35

# **EXECUTIVE SUMMARY**

Interest in work and self-sufficiency among welfare recipients was heightened among government officials, social service providers, academics, and the general public when Congress enacted the Personal Responsibility and Work Opportunity Reconciliation Act (PWRORA), which changed the focus of the welfare system from income support to work. PRWORA ended the six-decade-old Aid to Families with Dependent Children (AFDC) program and replaced it with Temporary Assistance for Needy Families (TANF). Notable features of the new law include a five-year federal lifetime time limit on cash assistance, stringent work requirements, and conversion of AFDC, an open-ended entitlement, to TANF, a capped block grant. One agency affected by welfare reform is the U.S. Department of Housing and Urban Development (HUD). According to one estimate, half of all families with children residing in public housing in the mid-1990s were AFDC recipients (Newman, 1999). The Department has long recognized that its clients face multiple barriers to self-sufficiency, and that housing assistance alone will not enable them to overcome those barriers. Consequently, HUD has encouraged housing providers to find ways to help residents obtain essential supportive services, including education and job training, job placement, child care, and transportation. HUD has a number of programs that could potentially help residents succeed in the new welfare environment. These programs, however, were designed prior to welfare reform and have not been examined systematically in the new welfare environment.

The transformation of the welfare system presented HUD with an important opportunity to conduct a preliminary assessment of its programs. As a result, HUD contracted with ICF Consulting and the Lewin Group to review the employment and training components of the following 13 HUD programs:

- □ Economic Development and Supportive Services (EDSS). The Economic Development and Supportive Services program provides competitive grants to PHAs to establish education, training, and supportive service programs.<sup>1</sup>
- □ **Family Investment Centers (FIC).** FIC provides families living in public housing with better access to education and employment opportunities by providing seed money to PHAs for the construction of service centers.<sup>2</sup>
- Family Self-Sufficiency (FSS). The Family Self-Sufficiency program aims to help residents of public housing and Section 8 rental certificate or voucher program participants become self-sufficient though education, training, and the provision of supportive services. The program emphasizes case management to help residents systematically plan their transition to self-sufficiency. It also provides financial incentives for participation by allowing participants to fund an escrow account in lieu of making increased rent payments when their incomes increase.
- □ **HOPE VI.** The goal of HOPE VI is to revitalize severely distressed public housing developments by simultaneously investing in buildings and residents. While the majority of

<sup>&</sup>lt;sup>1</sup> In 1999, HUD implemented the Resident Opportunities and Self-Sufficiency Program (ROSS), which consolidates and replaces EDSS and the Tenant Opportunity Program (TOP).

<sup>&</sup>lt;sup>2</sup> Although funding for the FIC program ended, the facilities constructed with the funds still operate.

HOPE VI funds are used for construction, rehabilitation, and provision of replacement housing, HOPE VI also provides limited funding for community and social services.

- □ **Jobs Plus.** Jobs Plus is a demonstration program constructed by HUD, HHS, Manpower Development Research Center and several foundations to test the impact of intensive and integrated employment and support services upon welfare recipients making the transition from welfare to work.
- □ **Moving to Opportunity (MTO).** Moving to Opportunity is a 10-year demonstration program that combines tenant-based rental assistance with housing counseling to help low-income families move to low-poverty neighborhoods.
- □ Neighborhood Networks. Neighborhood Networks is a HUD initiative that encourages the development of community-based resource and computer learning centers in privately owned, HUD-insured and -assisted housing.
- Section 3. Section 3 of the Housing and Urban Development Act of 1968 requires employment and other economic opportunities generated by HUD funding for housing and community development projects to be directed towards residents of low-income communities.
- □ **Step-Up.** Step-Up offers PHAs a programmatic framework to use in providing low-income individuals with skills training and a year of pre-apprenticeship on-the-job-training, thereby allowing residents to "step up" into registered apprenticeships.
- □ **Tenant Opportunity Program (TOP).** The Tenant Opportunity Program seeks to strengthen resident organizations and increase resident self-sufficiency by providing grants for organizational development, self-sufficiency activities, and conflict resolution.
- □ Youth Apprenticeship Program (YAP). On a demonstration basis, HUD funded eight PHAs to provide young residents of public and assisted housing with skills training, paid job experience, and ultimately job placement.
- □ **Youthbuild.** The Youthbuild program provides funding for education and on-the-job-training for 16- to 24-year-old public housing residents who have dropped out of school.
- Drug Elimination Grant. The Drug Elimination Grant Program is not a welfare-to-work initiative per se, but a program designed to help PHAs address problems of drug abuse and crime in public housing developments. Many PHAs use funding to develop employment programs as a means of fighting crime.

# I. STUDY OVERVIEW

This study, "Success in the New Welfare Environment," assessed HUD's efforts to help residents transition from welfare to work. We began this project by reviewing the literature on programs designed to help welfare recipients find jobs, remain in jobs, and advance in the labor market. Based on this review, we identified a number of program components that seemed to be

associated with significant positive outcomes, such as exits from welfare, increased rates of employment, or increased earnings. We grouped these elements into four broad categories:

- **Employment services** that have an employment or education and training focus. Specific components include:
  - Job search
  - Work experience
  - Subsidized employment
  - On-the-job training
  - Vocational training
  - Basic skills training
  - Post-secondary education
  - Assessment
  - Case management
- **Barrier amelioration services** that focus on:
  - Environmental barriers such as child care and transportation
  - Personal barriers including substance abuse, mental health problems, lack of social support, and domestic violence
- **Cash and in-kind supports** that augment earned income, including:
  - Earned income disregards
  - The Earned Income Tax Credit (EITC)
  - Food Stamps and other nutrition programs
  - Transitional Medicaid
- **Transitional supports** that address a range of workplace issues and include:
  - Life skills training
  - Peer support

These elements provided the framework that was used to identify and assess HUD's employment and training programs.

The research team conducted in-depth, on-site interviews with individuals from local housing authorities, employment and training agencies, welfare offices, and community-based organizations in five cities: Baltimore, Boston, Los Angeles, Milwaukee, and San Antonio.

Given both the small number of sites visited and the significant flexibility HUD provides its local operators in the design and operation of its programs, this study was not intended as an evaluation of any specific HUD program or particular housing authority. Rather, the purpose of the study was to evaluate HUD's employment and training strategy as a whole, exploring the program activities, both individually and as a group, and their relationship to the larger efforts in the community to help welfare recipients sustain employment. The study focuses on the following questions:

- 1. What are the specific service components of HUD employment and training programs?
- 2. To what extent are the practices consistent with the research on the effectiveness of those service approaches?

3. What is the nature of the linkages between the HUD programs and the programs being provided through the larger human services and employment and training systems in the community?

### II. SUMMARY OF FINDINGS

# A. HUD Programs are Generally Consistent with the Literature on Employment and Training Programs

HUD programs under review were generally consistent with promising welfare-to-work approaches identified in the literature and emphasized approaches and components adapted to a work-first environment. Most programs that offer basic skills education, training, or post-secondary education are related to the needs of the local labor market and are employment focused. Almost all programs stress job search or contain a job search component. However, two of the more effective program components, subsidized employment and on-the-job training, were not well represented in HUD programs implemented at the selected sites. Assessment and case management were among the most common program elements observed; however, while intuitively sensible, their connection to employment and self-sufficiency is not well established in published studies. Finally, almost all programs attempt to help residents overcome barriers to employment, such as child care, either directly through the provision of on-site services or through referrals.

# **HUD Program Components**

	Program Philosophy <sup>3</sup>			Program Component											
	Work-Focused	Ed/Training with Work-Focus	Ed/Training- Focused	Job Search	Work Experience	Subsidized Employment	On the Job Training	Vocational Training	Basic Skills Education	Post-Secondary Education	Assessment	Case Management	Barrier Removal	Cash/In-Kind Supports	Transitional Supports
EDSS <sup>4</sup>				А							В	В	В		
Family Investment Center				В	В			В	В		В	В	Α		В
Family Self-Sufficiency										А	А	Α	В	А	В
HOPE VI				В	В			В	В			В	Α		В
Jobs Plus				А	В			В	В	В	А	Α	А	А	В
Moving to Opportunity											Α	Α	В	Α	А
Neighborhood Networks				В				А			В	В	В		В
Section 3							В	В							
Step-Up				В		А		Α					В		
Tenant Opportunity Program											В	В	В		
Youth Apprenticeship Program				В			Α		А		В	В	В		В
Youthbuild				В				Α	А		В	В	В		В
Drug Elimination Grant											В		В		В

A: Primary Program Component (central feature of programs across all sites)

B: Secondary Program Component (lesser feature of program or site-specific use)

 $\sqrt{\cdot}$  Denotes program philosophy

<sup>&</sup>lt;sup>3</sup> Five programs, FIC, HOPE VI, MTO, TOP, and DEG, were not easily classified since they were not focused on work or education and training, but were vehicles to offer a combination of services to the resident population overall.

<sup>&</sup>lt;sup>4</sup> In two of the four sites, EDSS funds were used to support the Jobs Plus initiative. Freestanding, it appears as a heavily work-focused program. When combined with Jobs Plus, EDSS supports more comprehensive employment and training services.

#### 1) Most HUD programs are employment focused.

Three programs are employment focused: Section 3, Jobs Plus, and EDSS. While none excludes the option of training, the overarching focus of each program is work. Three training programs could be classified as education- and training-focused with a strong work element: Step-Up, YAP, and Youthbuild. They provide hands-on experience in a well-paying trade and often help place participants in jobs. Only two programs, FSS and Neighborhood Networks, could be classified as education- and training-focused initiatives.

The remaining HUD programs are not easily classified. These include FIC and HOPE VI, both of which provide space to housing authority and non-housing authority service providers. Services offered by these programs range from basic education to job search.

#### 2) Job search is a key program component.

Almost all programs offer some form of job search. For some, such as EDSS and Jobs Plus, job search is a central focus, and participants have access to both job-related and supportive services. In other programs, job search is one, albeit smaller, component of a larger intervention. Step-Up, YAP, and Youthbuild, for example, focus on training in specific fields, although participants often receive assistance finding jobs. Programs such as FIC and HOPE VI aim to provide a constellation of services to residents, and job search is one such activity. Other programs, such as FSS and MTO, refer clients interested in employment to other service providers. Neighborhood Networks also had job search activities in some sites.

#### 3) Training programs are generally linked to work.

HUD vocational training programs range from long-term (one or more years) efforts that seek to prepare participants for specific careers or trades to short-term or customized training. Most of these HUD programs train residents for jobs in demand. Youthbuild, YAP, and Step-Up train participants for construction-related jobs. Because housing authorities spend millions of dollars each year on construction and renovation, there is a demand for skilled labor. Only in YAP was there a guaranteed job; the others sought to place participants in permanent jobs, however. Other HUD programs, such as FIC, HOPE VI and Jobs Plus, incorporate short-term or customized training programs that focus on high-demand jobs such as child care and health services on a site-by-site basis. Neighborhood Networks focuses on computer skills.

#### 4) Education is often one part of a larger intervention.

No single HUD program focuses exclusively on basic skills education. Youthbuild, YAP, and Step-Up, for example, combine hands-on construction experience with basic education. Consistent with the literature, the educational component is viewed as a means of opening up future possibilities (e.g., jobs with increasing responsibility) and not as an end in itself. Other HUD programs, such as FIC, Jobs Plus, and HOPE VI, offer more self-contained basic education courses. In these cases, staff suggests that basic education is an important step toward self-sufficiency, one that will make a candidate more employable. Even in these instances, however, the basic education courses are one of several employment-focused activities that residents can select. FIC, Jobs Plus, and HOPE VI also offer job placement services and linkages to vocational training.

#### 5) Post-secondary education is one of the least common program elements.

Staff from most of the HUD programs noted that referrals to post-secondary institutions are possible under program guidelines, but that they are infrequent. FSS is the HUD program most likely to refer participants to post-secondary education. The Jobs Plus program in Baltimore also had agreements with local community colleges to offer customized training as needed. One factor for the infrequent post-secondary education referrals is the structure of the welfare programs in the five sites. Only the Maryland TANF program counted post-secondary education as a TANF work activity. In other sites, a TANF beneficiary would have to combine education with work.

#### 6) Subsidized employment and on-the-job training are less common.

These are relatively rare program elements, even though evaluations of subsidized employment and on-the-job-training programs found significant, positive impacts on employment and earnings. Three programs used these components. YAP and Section 3 used on-the-job training extensively. Subsidized employment, or using TANF grants to fund program services, was part of the Step-Up program. There appears to be an opportunity in some sites for housing authorities and welfare agencies to collaborate and use TANF funds to support HUD programs. In three of the five sites we observed subsidized employment and on-the-job training were allowable work activities under TANF rules. In Baltimore, for example, subsidized employment was used to fund a number of resident initiatives, in addition to Step-Up.

#### 7) Assessment can reveal barriers to employment.

Almost all HUD programs have an assessment component. In some cases, assessment is used to screen for program eligibility and possible barriers to participation, including substance abuse, skills deficits, and learning disabilities. In other programs, assessment involves gathering information on a client's educational attainment, work history, goals, and barriers to employment, with the aim of creating a service plan. The plan may focus on steps a resident could take to secure employment, or it may focus on overcoming barriers by referring the participant to GED classes, ESL, or other service providers.

#### 8) Case management is common, yet generally not focused on employment.

Case management is a central component of three HUD programs: FSS, Jobs Plus, and MTO. Other HUD programs incorporated case management on a site-by-site basis. For example, we observed case management services at some, but not all, HOPE VI, EDSS, Neighborhood Networks, Youthbuild, YAP, TOP, and FIC programs. Most often case management links a resident to supportive and/or educational services. To the extent that such services make a resident more employable, they are a positive step. However, they often fall short of a work-first orientation. Jobs Plus and EDSS had the strongest focus on employment.

#### 9) Direct provision of barrier amelioration services varies by site.

Assessment and case management are the primary methods for determining barriers to employment and helping residents access services. While all housing authorities visited provided referrals for services, a number of housing authorities provided on-site services or space for barrier amelioration providers, including child care and health services. Most often, on-site services were provided through the HOPE VI, Jobs Plus, and FIC programs. DEG is also used by housing authorities to co-locate supportive services at targeted housing developments. In addition, housing authorities offer barrier removal services outside of discrete HUD employment and training programs. In Milwaukee, for example, child care is available at all five family developments, and multiple developments have health care providers on site. In Baltimore, the housing authority is the largest provider of child care services in the city.

# B. Fewer HUD Employment and Training Programs Serve Section 8 Recipients

Families holding Section 8 vouchers and certificates have less access to employment-focused HUD programs since the three employment and training programs that target Section 8 recipients are generally not work-focused. FSS has historically promoted education and training as a means to self-sufficiency and has encouraged participants to obtain advanced degrees that will improve their long-term employability. MTO primarily focuses on providing opportunities for families to live in areas with low poverty, in the belief that these areas will offer greater job opportunities and generally improve the life chances of participants. MTO programs help families find and lease approved privately owned housing units. Post-move supports, including counseling, job development workshops, and referrals for social services and/or education and training are available but participation is not mandatory. With the implementation of local programs using the new welfare to work vouchers, there will be a much stronger focus on linking this tenant-based rental assistance with job search and placement programs.

An important issue to consider for programs that target these Section 8 recipients is the greater difficulty they may find in reaching these clients. Unlike residents in public housing developments, Section 8 recipients are often scattered throughout many neighborhoods. Welfare and labor department officials may find working with public housing developments a more attractive option because they provide access to a large number of TANF clients in one place. Individual private owners or managers of properties in which some residents receive tenant based Section 8 assistance may not have the capacity or inclination to establish linkages with welfare and labor agencies for those receiving TANF assistance. In these cases, access to PHA programs may need to be expanded to serve these residents.

For families receiving Section 8 project-based assistance in privately owned, HUD insured multifamily developments, HUD's Neighborhood Networks Initiative offers computer-based training and educational programs. However, the range of services offered is highly variable, and the number of developments that have Neighborhood Networks centers is very limited.

# C. TANF Resident Participation in HUD Programs is Uneven

Only one HUD program, Jobs Plus, primarily served residents on welfare in all sites observed. This is not surprising, since Jobs Plus was designed to help resident transition from welfare to work. Other programs, such as EDSS, FIC, FSS, HOPE VI, and YAP served TANF clients, though to a lesser extent.

One factor that hinders program participation for TANF clients is the welfare rules in the five sites. The welfare department in only one site, Baltimore, accepted participation in HUD programs as work activities. The welfare department in a second site, Los Angeles, allowed TANF recipients to participate in the housing authority's programs *only* if they were unable to find work during a mandatory three-week job club. In the remaining three sites, Boston,

Milwaukee, and San Antonio, welfare departments operated work-first programs that focused on employment at the earliest possible time. Consequently, acceptable work activities were those focused primarily on job search, and participation in education and training programs, including those offered by housing authorities, did not count.

Another factor that appears to limit participation in many HUD programs is the nature of the training. Step-Up, YAP, and Youthbuild are heavily focused on construction-related occupations. According to the staff at a number of housing authorities, many single mothers are not attracted to manual labor, while others find it difficult to arrange child care to deal with the long hours and travel requirements of construction work.

#### D. Programs that Target TANF Recipients Combine HUD Funding Streams and Harness Resources from the Broader Community

A number of the local efforts observed combined HUD program funding streams and community resources to create employment and training programs that are locally designed and targeted toward residents on welfare.

# 1) Jobs Plus pools resources from other agencies

The Jobs Plus required housing authorities to collaborate with welfare and labor agencies to design and fund a program to help TANF recipients transition from welfare to work. In Baltimore, for example, the \$200,000 Jobs Plus grant (awarded to each of the seven national Jobs Plus sites) was matched by \$250,000 from the housing authority, \$80,000 from the local labor department, \$100,000 from the Empowerment Zone, and \$50,000 from the local welfare department. The welfare department worked with the housing authority to craft program activities that will count toward the TANF work requirement. Thus, TANF recipients have a range of services available. The labor department is on site to help place participants in jobs. The on-site child care center facilitates the transition to work. Residents who are not job-ready, or want to improve their job skills, work with on-site case managers to access activities such as on-the-job training, work experience, and education programs.

# 2) Housing authorities combined HUD funding streams to create employment-focused programs tailored to community needs

Housing authorities also adapted existing HUD programs to serve TANF clients and other employment seekers. This often involved combining HUD funding streams or tapping into labor and welfare department resources. In Milwaukee, for example, the housing authority used EDSS and HOPE VI funds to finance resident employment coordinators. These staff persons are the point people for residents seeking work. They conduct the assessments and serve as case managers and job developers. In two of the five developments targeted by the program, the resident employment coordinators also work with staff from the county welfare department and the "W-2 agencies" that administer the TANF program and the city's one-stop job centers. Coordinators work in tandem with W-2 staff on job search assistance; county staff assists with Food Stamp, Medicaid, and child care issues.

In San Antonio, the housing authority worked with the labor and welfare departments to establish one-stop resource centers at nine housing developments. The need for a service

delivery system for the pending HOPE VI projects was the impetus for creating this model. While the housing authority's two HOPE VI developments are being constructed, residents of the public housing projects demolished to make way for the new projects have been temporarily relocated. The housing authority wanted to provide services to these residents in the interim, and developed a one-stop approach that became a model for a larger initiative. The housing authority partnered with the welfare and labor departments to create the Resident Opportunities and Achievement and Development (ROAD) Centers. It is providing space at nine developments for the Centers. In addition, FSS coordinators will be housed at the Centers to assist with counseling and supportive service referrals. The on-site welfare agency staff will provide eligibility determinations for TANF, Food Stamps, and Medicaid. The labor department will provide the employment services, including job readiness training, job placement, and post-employment services.

# 3) The Los Angeles housing authority's primary programs for TANF clients are outside of HUD programs under review

In Los Angeles, the primary employment programs for residents on TANF fall outside of the discrete HUD programs under review. The Welfare to Work program is funded by a competitive welfare-to-work grant from the U.S. Department of Labor. The local labor department provides space at its one-stop career centers for the housing authority's welfare-to-work case managers, and the Private Industry Council, which funds employment and training services in the community, provided funding to serve additional residents. The Welfare-to-Work program is designed to operate in a work-first environment. Staff assesses clients to determine career skills, develop a service plan, and assist with job search. TANF clients can receive their work-related services through the program so long as the welfare agency certifies eligibility. The local department of labor also funds "satellites" to the one-stop career centers at a number of housing developments.

# E. Strong Linkages Exist with Labor Agencies; Relationships with Welfare Departments could be Strengthened

We found varying levels of collaboration between housing authorities and the welfare and labor agencies in their communities. In all five sites, the housing authorities had a history of collaboration with the labor agencies. In Baltimore, for example, housing authority and labor department staff worked together to create the People Accessing Continued Employment (PACE) program at four developments to provide pre- and post-employment services to residents. In contrast, collaboration in Boston was more limited. In the past, the labor department simply funded outreach workers to inform residents about employment and training services in the community.

However, linkages with the welfare agencies appeared to be underdeveloped in many sites. While Baltimore and Milwaukee housing authorities have worked with welfare agencies in the past to design and administer programs to serve public and assisted housing residents, relationships and joint planning efforts have just begun in Los Angeles and San Antonio. In Boston, there is no coordination of employment services among high level housing authority and welfare department staff; although some coordination occurs between individual developments and local offices.

#### F. The Implementation of Financial Incentives for Residents Must be Resolved

HUD employs financial incentives to encourage public housing residents to work or seek training. The key HUD policy is contained in the 1998 Quality Housing and Work Responsibility Act (QHWRA), which replaced the 18-month mandatory income disregard with a 24-month disregard for residents whose income increases as a result of work or training. Under the new policy, housing authorities are required to disregard all increases in income from employment for an initial 12 months and 50 percent of increased income for a subsequent 12-month period. In addition to QHWRA, some individual housing authorities have similar financial incentives in place to encourage employment. Milwaukee, for example, disregards the first \$2,000 of earned income when a resident becomes employed and does not increase the monthly rent until the following annual review.

The negotiated rulemaking process, which will clarify QHWRA policy, was incomplete at the time of our site visits. We found significant confusion among PHA staff as to the nature of the earned income disregard spelled out under QHWRA and its impact on PHA revenue. At the Baltimore site, the housing authority had implemented a policy that matched the QHWRA guidelines, but in the other four sites, the housing authority staff we met with appeared to be aware of statutory change but unsure about how it will be implemented, in particular, whether and how operating subsidies would be adjusted to cover rent losses. In one case, the Los Angeles authority had asked HUD to review their income disregard policies and expressed concern about going too far without direction from HUD and clarification of how rent losses will be treated in the current year and in the future.

Nor did residents we interviewed appear to know the rules governing HUD's policies on income disregards, and we did not observe systematic efforts on the part of housing authorities to educate residents about the policy. In one site, Boston, a non-profit organization was working with the tenant organization in one development to educate residents about the pre-existing 18-month earnings disregard. This lack of clarity appears to be affecting PHA implementation of the policies, as well as resident benefits from participating in employment and training programs.

#### G. Housing Authorities Approach the Issue of Helping Residents Attain Self-Sufficiency Differently

The five housing authorities we observed also take varying approaches to helping residents attain self-sufficiency. On one end of the spectrum, the Baltimore and Los Angeles housing authorities take a comprehensive approach by providing a broad array of employment and training programs, as well as on-site support services such as child care. The Los Angeles housing authority buttresses this approach with a strong emphasis on using its own construction company, Kumbaya, and resident management corporation-owned businesses for renovation and modernization projects.

On the other end of the continuum, the Boston housing authority adopted a more limited and decentralized approach to service delivery, guided by three principles: residents should use services located off site; the housing authority's role is primarily as a referral agent to other service providers; and, individual tenant organizations should take the lead on addressing the needs of their residents.

Falling in the middle, the Milwaukee and San Antonio housing authorities operate in strong work-first welfare environments. Both take the position that housing providers should help

residents access employment and training services. In both sites, residents are often referred to outside providers. The Milwaukee housing authority also recognized that bringing community-based organizations on-site built important bridges to the community. Each family development has a number of on-site providers offering work supports ranging from child care to health services and food assistance. San Antonio is moving in the direction of on-site services. In summer 1999, the first ROAD Center opened at a housing development, and eight more will follow.

# III. CONCLUSIONS

HUD has a long-standing interest in promoting the self-sufficiency of residents in public and assisted housing. Crafting new policies at the national level, however, is difficult, because housing authorities have a great deal of flexibility in structuring employment and training services for their residents. While this flexibility is important to meet local needs, it makes it difficult to create standardized programs. Housing authorities determine whether services will be offered by their staff or others in the community; whether services will be offered on-site at developments or through referrals to community organizations; and even whether services will be provided at all. Housing authorities also decide whether and the degree to which they will work with local welfare and labor authorities.

Despite these variations, our observations and interviews with staff during the site visits suggest a number of steps HUD could take to strengthen employment and training services for residents. First, HUD could facilitate collaboration among housing authorities, welfare departments, and labor departments. HUD activities could include providing PHAs with technical assistance or training on collaborations, encouraging PHAs to focus on retention and advancement services for employed residents, including incentives for collaborations in HUD's assessment systems, and helping provide a stable funding platform for collaborations. Second, HUD could take steps to ensure that PHAs have implemented the financial incentives for working residents outlined in the 1998 Quality Housing and Work Responsibility Act. Finally, HUD could address the lack of participation in certain programs among residents on TANF by determining barriers to participation and adjusting programs accordingly. If this proves difficult, HUD could consider ways to target non-custodial parents for these programs.

#### A. PHA Employment and Training Activities for TANF Residents Should be Designed to Facilitate and Support Resident Participation in Welfare and Labor Funded Services

Housing authorities, and welfare and labor agencies, share a common goal of helping TANF recipients gain employment. A large proportion of non-elderly families in public and assisted housing receives TANF. Thus, housing authorities could lose rent revenue if residents lose their welfare benefits and fail to find or maintain employment. The welfare and labor agencies are responsible for helping TANF recipients access work and supportive services. All three agencies not only have a vested interest in helping residents find jobs but also in helping them stay employed. Clients who cannot maintain employment will quickly exhaust their lifetime TANF benefits.

Housing authority programs that target TANF recipients, such as Jobs Plus, the Welfare-to-Work program, the San Antonio one-stop Resident Opportunities for Achievement and Development (ROAD) Centers, and the resident employment coordinators, involve collaboration with welfare

and labor departments. In some cases, these agencies provide funding; in others, staff. In all cases, the housing authorities leverage resources from these outside agencies. This enabled them to conserve resources, which is important given their limited funding for employment and training and supportive services. Welfare and labor agencies, on the other hand, have considerable resources for these activities. In addition to TANF block grant funds, there are a host of financial resources in the community, including Workforce Investment Act (WIA) funds, Welfare-to-Work funds, and other state and local funds. The housing authority could assure access to a broad array of services for residents if it leveraged the resources available in the community. The question for housing authorities, then, is how they can augment and facilitate resident access to services provided by the welfare and labor agencies.

Housing authorities make attractive partners for their welfare and labor counterparts for a number of reasons. They provide access to a large number of TANF recipients in single locations, and may often be the first point of contact for residents seeking information about changes in the welfare system. As such, they are a resource for information on welfare and a source of referrals to welfare agencies and employment and training programs. Housing authorities can also offer space for on-site service providers, including welfare and labor staff. In addition, they can serve as "vendors" for welfare or labor department-funded employment and training services. Finally, housing authorities can provide opportunities at their developments for residents on TANF to meet their work requirements. For example, TANF recipients who cannot find unsubsidized employment could be offered community service jobs in the areas of maintenance, construction, or child care services. Housing authorities could also target available employment opportunities specifically to residents at risk of exhausting their benefits.

HUD can take meaningful steps to facilitate collaboration among housing authorities, welfare departments, and labor agencies.

# 1) HUD activities could support this strategy by providing appropriate written materials, training, and technical assistance to PHAs.

While housing authorities were selected for this study because each was operating several of HUD's employment and training initiatives under review, we also found variations in their level of knowledge of welfare policies, such as work requirements and acceptable activities, and in collaboration with welfare and labor agencies. To help housing authorities become more involved in helping residents meet work requirements, HUD could provide more information to PHAs on welfare reform, the roles of welfare and labor agencies, and the sources and uses of funds for employment and training services. This could include written materials on welfare policies at the state level, training on how to create partnerships with labor and welfare agencies, and technical assistance to help individual housing authorities develop strategies to fund employment and training activities. HUD could work with housing authorities that have established relationships with their welfare and labor counterparts to educate others about successful practices.

2) HUD could encourage PHAs to focus on retention and advancement services. TANF recipients who begin work for the first time often need help retaining their jobs and advancing in the labor market. Welfare departments are currently working to design initiatives in this area. One challenge is finding ways to encourage working clients to continue to access services and work supports. Welfare agencies often lose contact with clients after they begin working. Housing authorities are in an ideal position to address this service gap. Developments offer access to a large number of current and former TANF clients. Often space is available to co-locate staff from the welfare or labor departments. Using the Baltimore example, the People Accessing Continued Employment (PACE) program, designed by the housing authority and the city's labor department, provides pre-placement, post-placement, and job-replacement services on-site at a number of housing developments. In terms of advancement, efforts to help low-wage workers combine work and education could help them advance in the labor force. Housing developments could make space available for classes during non-work hours. Many developments also offer on-site child care.

HUD can give retention and advancement services visibility by encouraging housing authorities to meet with welfare agencies to discuss joint efforts. The housing authorities could receive funding from the TANF agencies to provide post-employment services. In addition, HUD can encourage housing authorities to participate in ongoing evaluations of retention and advancement services. The U.S. Department of Health and Human Service's Administration for Children and Families, for example, has made job retention and advancement a major priority. Housing authorities with retention programs (e.g. PACE) could join with their respective welfare agencies to develop strategies in this area.

# 3) HUD could include incentives for collaborative employment and training programs in the new PHAS and SEMAP assessment systems.

HUD has developed new assessment tools for public housing, the Public Housing Assessment System (PHAS), and for housing authority administration of tenant based Section 8 programs, the Section Eight Management Assessment Program (SEMAP). The PHAS assessment is a tool to determine the extent to which housing authorities will be deregulated and permitted wider discretion in management and operation of public housing. A high-performing authority may also be eligible for bonus points in certain grant competitions. The assessment system references self-sufficiency activities under criteria addressing resident satisfaction, but almost all of the rule language speaks to customer satisfaction surveys, not to employment and training. SEMAP attempts to implement a similar concept. By adding employment and training efforts as a specific factor in these assessment frameworks, HUD could increase the motivation of local agencies to expand their linkages to local welfare and labor agencies.

# 4) HUD and PHAs could consider establishing funding mechanisms to provide a stable platform for PHA collaboration with local welfare and labor agencies

Effective collaboration among the housing authorities and local welfare and labor agencies is time consuming for all involved. Staff time for planning and developing both informal relationships and formal cooperative agreements is a cost to housing authorities that, with a few exceptions, is paid from already over-subscribed operating and capital funds. In most cases, existing Resident Services budgets must cover what are essentially added administrative costs brought on by welfare reform. Some authorities may be able to cover a portion of these costs with Drug Elimination Grant funds, to the extent the activities are related to DEG program goals. Similarly, these activities may be funded as part of the cost of HUD-funded Family Self-Sufficiency coordinators in smaller PHAs.

HUD's Jobs Plus demonstration program is an example of a program explicitly designed to bring about collaboration between housing authorities and the key players in the new welfare system. Most of the funding needed to implement the demonstration was provided from foundations, with limited HUD funding. Authorities that received demonstration grants are expected to work closely with welfare and labor agencies to bring those agencies and their funded services to demonstration developments. While this model is still being evaluated, the collaboration envisioned makes sense intuitively, given the potential economic consequences of new TANF rules for both residents and housing authorities. Other agencies also are adopting this strategy. For example, the Department of Labor's Welfare to Work grant program (from which the housing authority in Los Angeles was awarded a competitive grant) emphasizes collaboration in planning and coordinating activities among numerous agencies involved in the program. HUD could recognize the administrative cost to housing authorities of creating collaborative relationships and find appropriate funding mechanisms to support these activities. At a minimum, these costs could be more explicitly recognized in public housing operating cost projections. In the future, the determination of administrative fees for the Section 8 voucher programs could also take account of the administrative costs of welfare-to-work related activities that benefit Section 8 recipients.

From a policy perspective, these public housing administrative costs could be addressed in the ongoing negotiated rulemaking on the new operating fund called for in QHWRA and funding included in HUD's future budgets. Along with the proposed funding of income disregards, this would provide important resources to help PHAs meet the QHWRA goals of transforming public housing developments into communities of residents who work.

#### B. HUD Could Move Aggressively to Publish Regulations on Financial Incentives to Work and Provide the Necessary Training and Technical Assistance to Help PHAs Implement the Policy

The Quality Housing and Work Responsibility Act (QHWRA) provides a number of tools to housing authorities to help them establish policies that will encourage unemployed residents to find work and working families to remain in public housing. These include policies that establish ceiling rents below current "economic rent" levels to attract working families, and give residents the choice of paying flat rents or income-based rents. The Act also outlines a new mandatory income disregard policy for residents whose income increases due to training or work. The new policy replaces HUD's previous 18-month income disregard policy with a two stage, 24-month disregard. PHAs must disregard all increased income from employment for an initial 12 months and 50 percent of increased income for a subsequent 12-month period. Our site visit interviews revealed that income disregards have been implemented unevenly, and that housing authority staff and residents alike are often confused about the rules and their impact on tenant rent payments.

#### C. Current HUD Programs that do not Serve TANF Recipients could be Reconfigured to Attract More Single Mothers or Redirected to Focus on Non-Custodial Parents.

TANF client participation in a number of HUD programs appears to be limited. These programs include YAP, Youthbuild, Step-Up, and Section 3. Housing authority staff described a variety of reasons for low participation, including the nature of construction work and the difficult hours required for single parents with children. HUD could help increase participation among single mothers by determining the barriers to participation and adapting the programs accordingly. If the long hours away from home often associated with construction are preventing participation, HUD could help housing authorities focus on jobs in the community that do not require a long commute, such as maintenance work. If single mothers find some aspects of construction work more appealing than others, programs could accommodate these interests.

If reconfiguring programs appears to be difficult or not possible, housing authorities could retarget existing programs toward non-custodial fathers. Site visits indicated that the majority of participants in the construction-related programs were men. Staff could not determine what proportion were non-custodial fathers. There was a general sense that many non-custodial fathers would be precluded from program participation because many are not on leases; others have criminal backgrounds. A policy of serving non-custodial fathers would also be consistent with the initiatives in the broader labor environment, including the Department of Labor's welfare-to-work grants and the Workforce Investment Act, that make serving this population a priority.

HUD could provide written materials and technical assistance on how to involve non-custodial fathers in its employment and training programs, addressing such issues such as past criminal activity and leases.

# INTRODUCTION

Interest in work and self-sufficiency among welfare recipients was heightened among government officials, social service providers, academics, and the general public when Congress enacted the Personal Responsibility and Work Opportunity Reconciliation Act (PWRORA), which changed the focus of the welfare system from income support to work. PRWORA ended the six-decade-old Aid to Families with Dependent Children (AFDC) program and replaced it with Temporary Assistance for Needy Families (TANF). Notable features of the new law include a five-year federal lifetime time limit on cash assistance, stringent work requirements, and conversion of AFDC, an open-ended entitlement, to TANF, a capped block grant. One agency affected by welfare reform is the U.S. Department of Housing and Urban Development (HUD). According to one estimate, half of all families with children residing in public housing in the mid-1990s were AFDC recipients (Newman, 1999). The Department has long recognized that its clients face multiple barriers to self-sufficiency, and that housing assistance alone will not enable them to overcome those barriers. Consequently, HUD has encouraged housing providers to find ways to help residents obtain essential supportive services, including education and job training, job placement, child care, and transportation. HUD has a number of programs that could potentially help residents succeed in the new welfare environment. These programs, however, were designed prior to welfare reform and have not been examined systematically in the new welfare environment.

The transformation of the welfare system presented HUD with an important opportunity to conduct a preliminary assessment of its programs. As a result, HUD contracted with ICF Consulting and the Lewin Group to review the employment and training components of the following 13 HUD employment and training programs:

- □ Economic Development and Supportive Services (EDSS)
- □ Family Investment Centers (FIC)
- □ Family Self-Sufficiency (FSS)
- □ HOPE VI
- Jobs Plus
- □ Moving to Opportunity (MTO)
- Neighborhood Networks
- □ Section 3
- □ Step-Up
- **D** Tenant Opportunity Program (TOP)
- □ Youth Apprenticeship Program (YAP)
- □ Youthbuild
- **Drug Elimination Grant**<sup>5</sup>
- D. Outline of Report

The report is organized as follows:

Chapter I describes the context for the study. It reviews the rules of the new welfare environment, as well as the implications of welfare reform for public housing authorities and residents in HUD-assisted housing.

<sup>&</sup>lt;sup>5</sup>DEG is listed last because it is not an employment and training-focused program.

Chapter II describes HUD's role in supporting the self-sufficiency of residents in public and assisted housing.

Chapter III examines the degree to which housing authorities work with local welfare and labor agencies.

Chapter IV first discusses the framework used to assess the HUD employment and training programs. It then reviews how HUD's employment and training programs compare to the elements of effective programs that were identified in the literature.

Chapter V summarizes the study's findings and proposes future steps for HUD. The Appendix contains the literature review.

### E. Overview of Methodology

Data collection activities focused on in-depth, on-site interviews with individuals from local housing authorities, employment and training agencies, welfare offices, and community-based organizations in five cities: Baltimore, Boston, Los Angeles, Milwaukee, and San Antonio. Given both the small number of sites visited and the significant flexibility HUD provides its local operators in the design and operation of its programs, this study was not intended as an evaluation of any specific HUD program or particular housing authority. Rather, the purpose of the study was to evaluate HUD's employment and training strategy on the whole, exploring the program activities individually, as a group, and in concert with the larger efforts in the community to help welfare recipients sustain employment. The study focuses on the following questions:

- (1) What are the specific service components of HUD employment and training programs?
- (2) To what extent are the practices consistent with the known research on the effectiveness of those service approaches?
- (3) What is the nature of the linkages between the HUD programs and the programs being provided through the larger human services and employment and training systems in the community?

We began the project by reviewing the evaluations of employment and training programs that targeted welfare beneficiaries. We focused on the following criteria of success: Did the program increase employment? Did the program increase wages? Did the program decrease welfare receipt and/or welfare payments? We summarized the common elements of these employment and training programs and assessed the degree to which HUD's programs share the components of successful employment and training programs.

A number of criteria were used to select five sites for review.

- Richness of HUD programs. We categorized HUD programs as core and secondary. Four programs, FSS, EDSS, HOPE VI, and Youthbuild, were designated as core programs because they share many of the components of the employment and training programs reviewed in the literature. We attempted to select sites that had all four programs, in addition to numerous secondary programs.
- □ *Welfare reform issues*. We selected sites with varying approaches to time limits, work requirements, and earnings disregards. These issues presented differing challenges for the HUD programs that serve HUD-assisted households.
- □ *Geographic distribution*. We included sites from the East (Boston and Baltimore), the South (San Antonio), the West (Los Angeles), and the Midwest (Milwaukee).

Key features of the five sites reviewed are summarized below.

# **Site Characteristics**

	Baltimore	Boston	Los Angeles	Milwaukee	San Antonio
Units Conven- tional					
P. Housing Section 8 <sup>6</sup>	17,100 11,300	14,440 8,400	8,800 44,000	4,000 4,800	9,400 9,500
Residents % Residents on TANF <sup>7</sup>	55,000 15%	26,000 24%	124,150 40%	23,150 9%	14,000 35%
Local Labor Department	Office of Economic Development	Mayor's Office of Jobs and Community Services and Private Industry Council	Community Development Department	Private Industry Council of Milwaukee County selected five "Wisconsin Works" (W-2) agencies to operate employment & training services	Alamo Workforce Development
TANF Agency	Department of Social Services	Department of Transitional Assistance	Department of Public Social Services	The five "W-2 Agencies" administer the TANF program	Texas Department of Human Services
HUD Programs Visited EDSS FIC FSS HOPE VI Jobs Plus MTO N. Networks Section 3 Step-Up	X X X X X X X X	X X X X X X X	X X X X X X X	X X X X	X X X
TOP YAP Youthbuild DEG	X X X	X X X X	X X X	X X X	X X

 <sup>&</sup>lt;sup>6</sup> Includes Section 8 vouchers and certificates. Still awaiting confirmation from Baltimore.
 <sup>7</sup> Proportion of non-elderly residents. Still awaiting confirmation from Baltimore.

## CHAPTER I: EMPLOYMENT IS THE HALLMARK OF THE NEW WELFARE ENVIRONMENT

HUD's employment and training programs operate in an environment in which welfare recipients are required to work. Under TANF's rules, simply finding a job will not be enough, however. Welfare recipients need to find jobs that enable them not only to leave welfare but to remain off welfare as well.

# A. PRWORA Fundamentally Changed Welfare Rules, Creating a Strong "Work First" Environment

On August 22, 1996, President Clinton signed PRWORA into law. The new welfare legislation repealed the 61-year-old AFDC program and replaced it with TANF.<sup>8</sup> The new law changed the nature of the welfare system by eliminating its entitlement status. The unlimited federal matching funds that characterized AFDC were replaced with fixed block grants. Each state's grant was based on its 1994 caseload. In addition, states must continue to spend their own funds. This "maintenance of effort" provision requires state spending to remain at or above 80 percent of 1994 spending levels. PRWORA also mandates a maximum five-year lifetime limit on federally funded benefits (although states can exempt up to 20 percent of their caseloads), and recipients are required to work after two years, or earlier at state option. Also, an increasing proportion of recipients must work each year: 25 percent of single parents were required to be engaged in work activities in FY 1997; by 2002, 50 percent must do so (U.S. House of Representatives, 1998).

Perhaps the most radical element of PRWORA is the devolution of the design, implementation, and management of welfare programs to the states. They determine the underlying philosophy of the program, the population that will be served, the size of the grant, the nature of mandatory activities, and the types of sanctions recipients face for not participating. States decide whom to exempt from work, when work starts, and how work is defined. States set asset limits (AFDC imposed a national limit), and can change the earned income that is disregarded for purposes of benefit calculation.

In essence, welfare reform created 50 state experiments. In some instances, decisions are made at the state level; in other cases, counties administer the TANF program. Employment and training services might be offered in-house by the welfare department; services might be contracted out to the state, county, or city department of labor, which already runs employment and training programs funded by the Job Training Partnership Act (JTPA) and other sources; or, welfare departments might engage non-profit organizations in the community. As a result, a housing authority that offers employment and training programs may have to work with numerous players and processes. These state variations also will affect how HUD employment and training programs operate at the local level.

# **B. State Characteristics and Policies**

The TANF rules that govern the five state welfare programs in the sites that were visited are outlined in Table 1.1. In accordance with our study's goal, we visited sites with a wide variation in welfare policies. The welfare grant for a three-person family (one parent and two children)

<sup>&</sup>lt;sup>8</sup> PRWORA also repealed the Emergency Assistance for Needy Families and the Job Opportunities and Basic Skills Training (JOBS) programs.

ranged from \$188 per month to \$673. Work requirements began immediately in some sites and after two years in others. Non-compliance with work requirements resulted in a full family sanction in two sites, and a partial grant reduction in three sites. We also observed variation in earned income disregard and transitional assistance policies.

As Table 1.2 indicates, each state has experienced a significant decline in its welfare caseload since 1994. Four states had declines that exceeded the national average of 49 percent. The cause of the declines is subject to debate. Some suggest that the decrease in caseloads is attributed to the 1996 welfare reform law and prior welfare reform initiatives (including state-level waivers from AFDC program rules) that required welfare recipients to work. Others claim that the decline in caseloads is attributed to the economic expansion experienced during the mid-1990s.<sup>9</sup>

<sup>&</sup>lt;sup>9</sup> Historically, welfare caseloads have tended to rise during economic recessions and decline during periods of economic recovery. The current welfare caseload reduction coincides with a reduction of the unemployment rate. Although the unemployment rate began to decrease in 1992, it still remained above its 1990 level of 5.6 percent until 1994. As the unemployment rate continued to decrease throughout 1995 and 1996, the number of AFDC recipients fell (Lewin Group, 1998).

	California	Maryland	Massachusetts	Texas	Wisconsin
Grant	Los Angeles: \$596 for a family of three.	\$328 for a family of three residing in public housing and \$388 for a family that does not reside in public housing.	Exempt families: <sup>10</sup> \$539 for family of three that resides in public housing and \$579 for a family that does not Non-exempt <sup>11</sup> families: \$525 and \$565, respectively.	\$188 for a family of three.	Determined by rung on employment ladder: \$673 for recipients in community service jobs; \$628 for those in W-2 Transitions. Recipients in unsubsidized employment or trial jobs receive wages.
Time Limits	18 to 24 month consecutive time limit (new applicants vs. current recipients); 60 months lifetime. Time limit applies to adults only; children can continue to receive a grant.	60 month time limit. Time limits apply to the adults only; children can receive a voucher or grant paid to a third party.	Receipt is limited to 24 out of a continuous 60- month period. Time limit applies to entire family.	12 months for recipients with GED/diploma and work experience. 24 months for recipients with GED/diploma or work experience. 36 months for recipients with no GED/diploma or work experience. Time limit applies to adults only.	24 month consecutive time limit on any rung of employment ladder; 60 months lifetime. Time limit applies to entire family.

Table 1.1: **State TANF Program Overview** 

<sup>&</sup>lt;sup>10</sup>Exempt from time limit and work requirement. <sup>11</sup>Non-exempt caseload divided between those who are mandated to participate in the work program and those who are not. Presence of a full-time school aged child determines work requirement.

	California	Maryland	Massachusetts	Texas	Wisconsin
Work Reg.	Recipients are required to accept any legal job and must work 32 hours per week (two-parent families are required to work 35 hours/week).	Recipients are required to work after 24 months for 25 hours per week.	After 60 days, non-exempt recipients must be in subsidized or unsubsidized jobs or participate in community service for 20 hours per week.	There is a 25-hour per week requirement for single parents and a 35-hour per week requirement for two- parent participants.	Adults are required to participate in program activities or work immediately; time requirements are generally 40 hours per week.
Sanctions	The adults are removed from the grant in instances of non-compliance (3 months for first sanction; 6 months for subsequent sanctions).	When the recipient is sanctioned, the grant is terminated until compliance (full family sanction).	Removal of adult's needs for first sanction; benefits are terminated in instances of continued non-compliance. Sanctions are lifted upon compliance.	Benefits are terminated in instances of non- compliance. Compliance reinstates the grant after the first sanction; repeated sanctions require 30 days of compliance before benefits are restored.	Recipient's benefits are tied to the number of hours that they work/participate and are decreased by the minimum wage for every hour of non- participation.
Earnings	\$225 plus 50% of earned income and unearned disability income are disregarded.	20% of earnings disregard plus \$200/month in allowable child care expenses.	\$30 plus 50% for non- exempt recipients; \$30 plus 33% for exempt recipients.	\$90 deduction for each working household member plus standard \$30 and 33%.	Grant amounts are based upon hours of work/program participation; there are no earnings disregards.
Transitional Asst.	12 months of transitional Medicaid assistance is available; 24 months of transitional child care is available.	Transitional Medicaid and child care assistance are both available for 12 months.	Transitional Medicaid and child care assistance are both available for 12 months.	18 months of transitional Medicaid assistance; 12 months of transitional child care.	Former recipients may receive 12 months of transitional Medicaid and up to 12 months of transitional child care.

# Table 1.1 (continued) State TANF Program Overview

Source: Site visits.

State	January 1994	January 1995	January 1996	January 1997	January 1998	March 1999	Percent Decrease ('94-99)
California	2,621,383	2,692,202	2,648,772	2,476,564	2,144,495	1,818,197	30.6%
Maryland	219,863	227,887	207,800	169,723	130,196	89,003	59.5%
Massachu-	311,732	286,175	242,572	214,014	181,729	151,592	51.4%
setts							
Texas	796,348	765,460	714,523	626,617	439,824	313,823	60.6%
Wisconsin	230,621	214,404	184,209	132,383	44,630	28,863	87.5%
U.S. Total	14,275,877	13,930,953	12,876,661	11,423,007	9,131,716	7,334,976	48.6%

Table 1.2 Number of Individuals on AFDC/TANF, by State

Source: U.S. Department of Health and Human Services, Administration for Children and Families: August, 1999.

# C. Impact of Welfare Reform on Public Housing Residents or Families Receiving Section 8 Assistance

Public assistance is the primary source of income for almost one half of all families with children living in public housing or receiving Section 8 tenant based assistance. (HUD Research Fact Sheet, 1998). Families in assisted housing also comprise a significant proportion of the AFDC/TANF caseloads. A 1998 GAO study found that 29 percent of all TANF recipients nationwide also received housing assistance (GAO, 1998). While there is considerable state to state variation in this pattern, the overlap of housing assistance and welfare receipt has increased on a national basis since 1981 (Newman, 1999). Research on the characteristics of households that receive welfare and live in public housing shows they have been on welfare longer, on average, than other welfare recipients, and thus may be among the harder to serve population of welfare recipients (Newman & Harkness, 1999). On average, households receiving rental assistance also have extremely low incomes, approximately 21% of the area median income (HUD, 1998).

Welfare reform is having notable impacts on Public Housing Authorities (PHAs) as well as on welfare recipients, and these impacts are helping to transform public housing communities. The immediate reaction to welfare reform by welfare recipients is fear of the loss of their benefits if they are unable to find a job before their TANF benefits expire. The process of looking for a job and going to work may also be frightening for those with little work experience. Many also face real problems getting to employment locations and concerns about child care while they are away from home.

Once employed, many residents will need post-employment support and counseling to help them remain employed, advance in their current jobs, or find better ones. In many cases, entry-level jobs do not provide enough pay or benefits to replace all of the benefits received while on welfare, especially if child care and transportation services must be purchased. The transition to a stable work situation may involve a number of job changes, and result in a cycling in and out of the welfare system over a period of time.

These issues are largely the same whether a TANF recipient receives housing assistance or not; however, to the extent that assisted housing residents have been on welfare longer and have greater dependency on welfare as a source of household income, they may require more intensive training and services to make them job ready. Unlike other recipients, however, assisted housing residents do not need to fear a loss of housing or housing assistance if they make a good faith effort to find work and comply with other TANF rules. This is a great advantage and could make it easier for this group of TANF recipients to weather the challenges inherent in making the transition to work.

# D. Impact on Public Housing Authorities

The most obvious challenge to PHAs from welfare reform is the potential rent loss that may result if residents lose some or all of their income as a result of changes in welfare policy. The loss of welfare benefits as a result of time limits or sanctions, and the effect of income disregards for welfare recipients that obtain employment, may significantly impact project revenues for some period of time. State welfare time limits and sanctions vary, but in almost every case, there is the potential that recipients that are not exempt from the rules will lose all or a part of their benefits if they are not employed within a specified period of time.

Public housing policy generally requires tenant rents to be reduced if income declines, so that rents do not exceed 30 percent of income. However, in an effort to be consistent with welfare reform, PHAs are prohibited from reducing rents when benefits are lost for failure to comply with welfare rules. For residents who make a good faith effort, but fail to find employment, their rent will be reduced if their benefits are reduced or eliminated due to time limits.

Given the large number of welfare recipients in public housing for which these benefits are the primary source of income, individual developments are likely to lose some rental income. A 1998 GAO review found wide variation in the conclusions of 13 studies examining the likely impact of welfare reform on HUD subsidies. For example, one study estimated that annual subsidies overall would need to increase by over 40 percent, while another projected a 20 percent reduction in the subsidy needs for a particular agency (GAO, 1998).

A HUD study of eight housing agencies found an approximate range of plus or minus 20 percent in the subsidy needed to offset the impact of welfare reform. The study also found that nonexempt welfare recipients<sup>12</sup> contribute as much as 30 percent of total PHA rent revenue, and in some cases almost two-thirds of these residents must find full-time work if tenant rent levels are to remain stable (HUD, 1998). Other studies also suggest that tenant income may rise considerably. This may be especially true in markets in which there are labor shortages, where employers are eager to train and hire welfare recipients at wages above an entry level rate. This financial uncertainty for PHAs may be compounded by uncertainty over the funding of income disregards for tenants participating in a training program or employment by a previously unemployed household member. HUD's former mandatory policy to disregard earnings for up to 18 months was not in widespread use. Staff from several housing authorities suggested that there was concern that the resulting loss of income would not be recoverable from HUD subsidies. QHWRA replaced that policy with a mandatory disregard of all increased income from employment for an initial 12 months and 50 percent of increased income for a subsequent 12-month period. Under the new accounting rules for PHAs contained in QHWRA, it is not entirely clear how PHAs will be compensated for implementing the income disregard policies.

<sup>&</sup>lt;sup>12</sup> The study defines "non-exempt" as recipients not exempt from the welfare work requirements.

Negotiated rulemaking is expected to address this issue as it applies to public housing units, but until that is complete, PHAs cannot fully gauge the financial effect of these "hold harmless" policies.

Uncertainty also exists with respect to the Section 8 vouchers and certificates administered by PHAs. If more subsidy is needed per unit to offset loss of income from tenant rent payment due to termination of welfare, PHAs may need to draw upon Section 8 reserves, if they exist, or face a possible reduction in the overall number of units that can receive assistance within the amounts provided under the existing housing assistance contract.

PHAs also face continued budget cutting pressures by Congress. All housing programs are being scrutinized in an effort to reduce subsidy costs. PHAs will be able to modify their tenant selection policies to give preferences to working families with higher incomes. Ceiling rents are one example of tools now available to PHAs to maintain existing working families in public housing as their income grows, by setting a limit on the amount a tenant's rent can be increased. While these are important steps that should ultimately result in the greater financial stability of developments, real increases in the share of rental income from tenants will occur slowly, as the new occupancy policies are applied over time.

# E. Impact on Housing Communities

Welfare reform should also impact the way public housing communities operate. As a result of stringent work requirements, many more PHAs can be expected to become more involved in counseling, case management, and employment and training activities to help residents prepare for and find employment. In some cases, welfare services may be co-located with employment and training programs at housing developments, where the concentration of TANF households in public and assisted housing developments could make it easier for welfare agencies to reach a large proportion of their caseload. In addition, some housing authorities are already attempting to provide more job opportunities for residents in maintenance and modernization, using resident organizations as contractors, and in resident service positions supporting welfare to work activities. Many provide training and apprenticeship opportunities for these jobs, or to help residents start their own businesses, such as in-home child care.

These efforts may provide new energy to change the environment of many housing developments. Negative impacts on housing communities could result from tenants who do not comply with the rules of the new welfare environment and who will consequently face dire financial circumstances. Positive impacts on housing communities could result from more adults and youth being engaged in structured activities both on site and off, and a wide array of service providers interacting with residents. Some residents are also taking vocational classes at local community colleges or high schools. These relationships and activities may, over time, combine to break down the barriers that have made many developments isolated from the larger community. While the outcome for welfare reform will not be known for some time, the integration of services should help many households receiving housing assistance increase their ability to become self sufficient, and in turn enhance their life choices. At the same time, changes in housing policy to encourage occupancy by residents with a greater mix of income levels should lead to more vibrant and economically viable public housing developments.

# CHAPTER II: THE IMPORTANT ROLE OF HUD IN SUPPORTING TRANSITION TO WORK

# A. Program History

When HUD officials articulate the view that public and assisted housing should be a platform for families to achieve self-sufficiency, they are bringing a much sharper and stronger focus to HUD's role and the role of housing assistance in fostering self-sufficiency. Public housing was initially intended to provide a pathway for working families to advance up the economic ladder. Over the years, the profile of those living in public or assisted housing changed dramatically, as a result of housing policies that targeted limited resources almost exclusively to the poor and unemployed; in public housing, this tended to create a de-facto holding area, in which families had limited access to the benefits of the larger society.

HUD has long recognized the important role that housing managers and owners can play in helping residents obtain services that will equip them for work. Beginning in 1984, with Project Self-Sufficiency, HUD initiated a series of demonstrations and programs to encourage and support action by PHAs and private landlords to help residents improve their education and their chances for employment. Over the next 15 years, Operation Bootstrap, Family Self-Sufficiency and similar initiatives tested various approaches and inducements to get PHAs and other housing providers involved in self-sufficiency efforts, and in forming linkages with the larger social service marketplace. Moreover, HUD's interest in welfare is not limited to the public housing venue, but includes other economic and community development programs such as Community Development Block Grants and Empowerment Zones/Enterprise Communities.

HUD has also collaborated in funding a number of research projects on welfare-to-work issues. Many of these initiatives were designed, often in partnership with foundations and others, to examine the dynamics of welfare reform as it affects housing residents, and the effect of housing assistance in supporting the transition to self-sufficiency.

As this study was undertaken, there were at least 12 programs or demonstrations in operation that HUD characterized as "welfare to work" initiatives. Each of the programs has a somewhat unique focus and characteristics, but all are intended to encourage housing providers to provide residents with the information, access to employment services and other transition services they will need to join the world of work successfully.

# 1) Program Framework and Current Programs

HUD requirements for each program are intentionally broad – that the local program address basic goals and objectives, and that a plan of services be developed to meet identified needs. In this way HUD sought to give PHAs great flexibility to design programs that fit within the overall welfare and economic environments in their localities.

Underlying HUD's thinking was the belief that housing developments should be closely linked to existing social service networks. In the past, service providers' targeting strategies may have overlooked public housing residents. In the new welfare environment, a wealth of funding is available to provide services, and the work first mandate upon states should change the view of service providers toward assisted housing residents.

Service providers should recognize the leverage that this linkage will provide to help them meet their new welfare responsibilities. PHAs are logical partners to TANF agencies and job-training agencies, because of their ability to access the thousands of households on welfare rolls that are

living in public or assisted housing. Moreover, the housing developments and housing authority offices have the physical space and facilities for on-site outreach activities and service delivery. The following is a brief description of the HUD welfare-to-work initiatives examined in this study of the role of HUD programs. These descriptions identify the broad characteristics that seem most relevant to the current welfare environment. The actual design and the implementation in each locality depends heavily on the philosophy of the PHA with respect to their role in welfare reform, the state and local welfare policies and programs in effect, and the collaboration between local housing authorities and welfare and service providers.

a. Economic Development and Supportive Services

The Economic Development and Supportive Services (EDSS) program enacted in 1966 provides grants to PHAs to help residents obtain education, training, and social services needed to obtain and maintain employment.<sup>13</sup> Eligible activities include: job training, job search assistance, job development and placement, and continued follow-up assistance; literacy training, GED, ESL, junior college and trade school assistance; supportive services, such as outreach and referral services for substance and alcohol abuse treatment and counseling or mental health counseling; child care; transportation assistance; and case management. A majority of the proposed activities must be administered either on-site or at community facilities within easy access to the housing development.

#### b. Family Investment Centers

The Family Investment Center (FIC) program provided grants to PHAs for the construction of resource centers that could be used to provide services that would help families access basic education and training needed to become more self-sufficient.<sup>14</sup> FIC funds could be used for: renovation or conversion of vacant dwelling units to create common areas to accommodate the provision of supportive services; the acquisition, construction, or renovation of facilities located near the premises of one or more housing developments to accommodate the provision of supportive services; the provision of up to 15 percent of the total cost of supportive services; or the employment of service coordinators. Eligible supportive services include: childcare; job training and placement; computer skills training; education (including GED, ESL, and post-secondary); and case management. New funding for the program ended in 1995, although some grants are still active.

c. Family Self Sufficiency

The Family Self-Sufficiency (FSS) program was created to help residents of public housing and recipients of tenant-based Section 8 assistance become self-sufficient through education, training, and the provision of supportive services, including case management. A participant must sign a contract with the PHA specifying what steps both the family and the PHA will take to promote the family's financial independence. Incentives for resident participation in the program include the ability to fund an escrow account in lieu of making higher rent payments as a result of increases in earned income during the term of the contract. During participation in the program, escrow funds can be used to help pay for college tuition or for other purposes consistent with the contract of participation. After completing the program, a participant can use

<sup>&</sup>lt;sup>13</sup> In 1999, HUD implemented the Resident Opportunities and Self-Sufficiency Program (ROSS), which consolidates and replaces EDSS and TOP.

<sup>&</sup>lt;sup>14</sup> Although FICs are no longer funded, the facilities constructed with the grants continue to operate.

funds for any purpose. The PHA must have a HUD-approved Action Plan developed in consultation with a local Program Coordinating Committee (PCC) comprised of representatives of local government, job training and employment agencies, local welfare agencies, educational institutions, child care providers, non-profit service providers, and businesses. *d. HOPE VI* 

The goal of HOPE VI is to revitalize severely distressed public housing developments and to create economically mixed neighborhoods by replacing aging, densely populated and economically isolated high rise developments with lower density, newly constructed single family and rental housing units. In addition to a development plan that meets these goals, each grantee must submit a Community and Supportive Services Plan (CSS) that identifies resident service needs and a plan of action for meeting these needs through community service providers and partners. Successful applications demonstrate coordination with welfare reform requirements and incorporate opportunities for resident self-sufficiency, particularly for persons enrolled in welfare reform goals; resident training, self-motivation, employment, and education; opportunities for economic and retail development at or near the public housing site, as appropriate; commitments by service providers; and potential employment opportunities for residents who complete community and supportive service training.

#### e. Jobs Plus

Jobs Plus is a demonstration program constructed by HUD, HHS, Manpower Demonstration Research Corporation and several foundations to test the impact of intensive and integrated employment and support services upon the transition of welfare recipients to work. The demonstration, which is operating in seven cities at eight public housing sites, attempts to measure the effect on employment, income, and welfare receipt of resident participation in a collaborative program that strongly emphasizes rapid employment, provides targeted employment training and education and creates an environment that supports work. A local collaborative of Public Housing officials, resident organizations, TANF and employment and training agencies will design and implement local programs to "saturate" the sites with services to support rapid employment and transition to self-sufficiency.

f. Moving to Opportunity

Moving to Opportunity (MTO) is a 10-year HUD demonstration program in five cities that combines tenant-based rental assistance with housing counseling to help very low-income families move from areas of high concentration of poverty to low-poverty neighborhoods. MTO will test the impact of housing counseling and other assistance on the housing choices of Section 8 households and the long-term effects of moving to low-poverty neighborhoods on the housing, employment, and educational achievements of the assisted households.

#### g. Neighborhood Networks

The Neighborhood Networks initiative encourages the development of community-based resource and computer learning centers in privately owned HUD-insured and -assisted housing. The goal is to increase residents' employment opportunities, improve educational performance of children, empower residents, and decrease residents' dependency on federal assistance. Neighborhood Networks is not a grant program; it encourages centers to become self-sustaining through partnerships, business opportunities and other income-generating options. While each

program is unique, Centers usually offer computer access and training, Internet access, job readiness support, GED certification, health care and social services, adult education classes, and youth services.

#### h. Section 3

Pursuant to Section 3 of the Housing and Urban Development Act of 1968, recipients of HUD funding must provide, to the greatest extent feasible, low- and very-low income persons living in or near public or assisted housing with a portion of the training, employment, and contracting opportunities that are generated by HUD financial assistance. Activities supported by assistance given to housing authorities for the construction, modernization, and/or operation of public housing developments are subject to Section 3 requirements. Section 3 applies to a broad range of activities within a covered project, including architectural services, construction, landscaping, maintenance and repair, purchasing, word processing, accounting, catering, and others. While these positions would ideally result in permanent employment for residents, there are limitations to project- or contract-related work. Although the regulations specify standards for technical compliance, methods for linking residents with jobs and contracts are not prescribed. Because there is no funding for Section 3, PHAs must use other sources of funds or rely on partnerships with service providers to establish training and supportive service programs.

i. Step-Up

The Step-Up program goal is to move people from welfare and unemployment to self-sufficiency through pre-apprenticeship training that will enable participants to qualify for apprenticeship programs. In addition, participants earn income as they train, generally over a one-year period. At the national level, the Step-Up program provides a framework within which local programs are designed and implemented. Local programs create partnerships with the U.S. Department of Labor or state approved apprenticeship programs for on the job training, and with local providers of education to provide supplemental classroom training and GED programs. Funding comes from a variety of federal, state, local, private and non-profit sources. Step-Up has historically focused on building maintenance and construction, but some programs have started to initiate training in day care as well. In general, participants work for an employer as a pre-apprentice for a maximum of one year and spend approximately half of their time receiving classroom instruction to earn their GED or high school diploma.

j. Tenant Opportunity Program

The Tenant Opportunity Program (TOP) seeks to strengthen Resident Organizations (ROs) by providing grants for organizational development, self-sufficiency activities, and conflict resolution.<sup>15</sup> HUD awards TOP grants in three categories: Economic self-sufficiency, organizational development, and mediation. Economic self-sufficiency (ESS) grants provide assistance to site-based ROs to establish programs that help move welfare-dependent families to work. Eligible ESS grant activities include: feasibility studies to determine training and social service needs; training in management-related trade skills and computer skills; coordination of support services (including child care, educational services, and health care outreach); business development training; and technical assistance for job training and placement programs. Organizational development grants provide assistance to help residents establish or enhance the

<sup>&</sup>lt;sup>15</sup> ROSS consolidated and replaced TOP and EDSS.

capacity of an RO. Mediation grants help provide training for conflict resolution between ROs and their housing authorities.

k. Youth Apprenticeship Program

The Youth Apprenticeship Program (YAP) aims to be a comprehensive job linkage program, providing residents with education and skills training, job experience, and ultimately, job placement. Grants were awarded to eight housing authorities in 1995. The YAP utilizes a multistep approach to assist young residents (ages 16 to 30) of public and HUD-assisted housing to achieve self-sufficiency including paid community service, pre-apprenticeship training and apprenticeship placements. Each YAP site must partner with a minimum of three organizations: a youth corps to operate the initial pre-employment component of the program; a local labor organization to facilitate participants' entry into apprenticeships; and a multi-employer organization to provide "assured" employment of apprentices for a minimum of 30 months. Additionally, HUD encourages sites to utilize local service providers for recruitment assistance, case management, childcare referrals and other essential services.

l. Youthbuild

Although not specifically focused on welfare recipients, Youthbuild provides low-income and very low-income young adults between the ages of 16 and 24 with opportunities to obtain education, employment skills, and on-site construction work experience as a service to their communities and a means to help residents achieve self-sufficiency. Participants are typically high school dropouts, although a portion of the participants may be high school graduates. The national Youthbuild office requires three components when it administers grants to local programs. First, a Youthbuild program must emphasize educational and job training services. Educational services generally focus on secondary education so that the participants may earn a GED. Job training services include topics such as how to look for a job and how to fill out an application. Second, Youthbuild programs must provide on-site training on the rehabilitation or construction of affordable housing. Finally, support services such as leadership training, substance abuse prevention counseling, and transportation assistance must be provided.

m. Drug Elimination Grant

The Drug Elimination Grant Program is not a welfare-to-work initiative per se, but a program designed to help PHAs address problems of drug abuse and crime in public housing developments. Grantees have been able to fund program activities and services that were reasonably related to drug abuse and crime prevention, including job training and education. The flexibility has enabled PHAs to use grant funds for resident services and programs that benefit welfare recipients and their families.

# B. New Legislative Mandates and Initiatives Reflect Changing Environment

As the momentum of welfare reform has grown, Congress and the Executive Branch have incorporated the philosophy of work first in a range of legislation and programs. The Quality Housing and Work Responsibility Act of 1998, a comprehensive public housing reform bill, and the provision of Welfare to Work Section 8 vouchers are among the most significant legislation affecting public housing authorities. The following are brief descriptions of this legislation and HUD's continued effort to refine its welfare to work programs.

## 1) Quality Housing and Work Responsibility Act

The Quality Housing and Work Responsibility Act (QHWRA) establishes a number of important public housing policies to help housing residents become self-sufficient and strengthen the role and responsibility of PHAs in addressing poverty and resident self-sufficiency issues. It provides important tools to housing agencies to establish policies that will encourage unemployed residents to find work and working families to remain in public housing. PHAs are also explicitly encouraged to develop cooperative arrangements with welfare agencies to target services to assisted housing residents, including bringing employment and training services on-site to facilitate participation by residents.

QHWRA gives additional discretion to PHAs regarding eligibility standards and rent policies for public housing that will expand the income mix in the program and help ease the transition to work for TANF recipients.

The additional discretion allows PHAs to:

- Establish eligibility policies that give preferences to families on waiting lists that are making the transition from welfare to work; and expand policies that allow for occupancy by higher income households, while meeting new income targeting guidelines;
- Establish ceiling rents below current "economic rent" levels, to attract and retain working families;
- Give public housing residents a choice of paying flat rents or income-based rents in order to eliminate the work disincentive for working families and encourage them to stay in public housing; and
- Implement reasonable minimum rent policies to help preserve the financial stability of properties.

As discussed earlier, QHWRA replaced HUD's existing income disregard policy, which only applied to public housing residents, with a new 24-month policy that applies to public housing and may be applied to Section 8 programs, if funding is appropriated by Congress. Residents currently receiving TANF assistance or those that received assistance within the prior six months are eligible for the disregard, as are households whose total income has increased as a result of employment of a previously unemployed family member or as a result of participation in a family self-sufficiency or job training program. PHAs continue to have the authority to establish their own earnings disregard policies for public housing tenants. The new Act also gives PHAs the authority to disregard other types of income beyond earnings, such as pensions and public assistance.

QHWRA also changes current policy for the Family Self-Sufficiency (FSS) program, eliminating the mandatory features of the existing FSS program for new incremental Section 8 units, and reducing the existing participation requirement by the number of families that successfully complete their contracts of participation. Limited HUD funding for program services remains an issue of concern to PHAs that are not eligible for or have not received competitive grants for service coordination (Center for Budget and Policy Priorities, 1998).

## 2) FY 2000 HUD Budget - Welfare to Work Vouchers

HUD's 1999 Welfare to Work Voucher initiative is one of the most exciting new opportunities to help welfare recipients who need housing assistance to obtain employment. Families eligible for, currently receiving, or who have left public assistance in the last two years are eligible, except those currently holding Section 8 vouchers or certificates. HUD's goal is to use the vouchers, in

conjunction with other social service funding, as a means of helping welfare recipients to seek or maintain employment.

Funding for 50,000 vouchers has been approved to help welfare families find or retain housing in areas with high job opportunities or with transportation systems that link to such employment areas. The vouchers were awarded on a competitive basis to 121 state and local housing agencies that demonstrated that the vouchers will be used as part of a coordinated plan of employment and training services. The plans were developed in cooperation with Department of Labor Welfare to Work administering agencies, and have the support of TANF administrative agencies and other service agencies.

## 3) Resident Opportunities and Self-Sufficiency Program

HUD has continued to refine its welfare-to-work initiatives to bring them into alignment with the current welfare environment, with Jobs Plus as a primary example. In 1999, HUD implemented the Resident Opportunities and Self-Sufficiency Program (ROSS) which consolidates and replaces EDSS, TOP and Public Housing Service Coordinators, and provides grants to public housing agencies for supportive services, resident empowerment activities, and employment and training services. ROSS grants are intended to strengthen resident organizations' capacity to increase resident involvement in their housing developments, and to help residents obtain employment training or to set up their own businesses. Funds may also be used for facility improvement for centers providing self-sufficiency related services and activities. Grants will be targeted to properties where more than half of the residents receive TANF benefits or are otherwise affected by welfare reform, and to programs with significant welfare-to-work focus and linkages to TANF agencies and other service providers.

## C. Creating Successful Housing and Welfare Strategies at the Local Level

Housing authorities have a great deal of discretion in the provision of employment and training initiatives for residents. As noted earlier, HUD requirements for each program are purposely simple to allow housing authorities the flexibility to shape programs to local needs. PHAs decide whether services will be offered on site or off site; whether their staff or others from the community will provide services; and whether activities are determined centrally or by individual housing developments. PHAs can choose whether to operate their own employment and training programs or to help residents obtain services from other providers in the community.

Although PHAs have flexibility in service design, state and local welfare rules provide the platform upon which welfare-to-work service programs are built. While the philosophy of federal welfare reform is to move recipients into employment as rapidly as possible, state discretion in setting policies with regard to time limits and sanctions are the determinant factors in program design. State and local welfare departments also have the option of offering employment and training programs in-house or utilizing other services in the community, such as labor department programs.

The degree to which PHAs are involved in the larger employment and training environment varies considerably. PHAs may help with information dissemination and referrals, may provide facilities for centralized delivery of services, may be a vendor of employment and training services, may provide direct services such as adult education and child care, and may be a source of employment through creation of Section 3 jobs. Participation in PHA programs may or may not count toward a TANF recipient's work requirement.

The five housing authorities we visited had unique philosophies regarding the role of housing providers in the provision of employment and training services.

#### 1) Baltimore

The housing authority takes a comprehensive approach to meeting the needs of its economically disadvantaged residents. While advocating a "work first" approach to welfare, the housing authority recognizes that residents need jobs that can sustain them. Its role is not only to provide stability through the provision of housing, but also to help residents achieve self-sufficiency by offering a broad array of employment and training programs as well as supportive services, such as on-site child care, at its developments. The job placement programs were designed based on analyses conducted by local universities that determined which occupations were declining and expanding locally.

The housing authority is part of the Department of Housing and Community Development, whose commissioner serves in the Mayor's cabinet. Housing authority staff has a long history of joint collaboration with their counterparts at the welfare and labor agencies. Housing authority staff negotiated with the welfare department to count HUD employment and training programs toward the TANF work requirement. Under Maryland law, welfare recipients must work after 24 months of benefit receipt. Work is defined broadly to include 25 hours per week of unsubsidized employment, job search, subsidized employment, work experience, on-the job training, and education or training directly related to employment. Housing authority staff has also worked on joint-planning efforts with the local labor agency, including the People Accessing Continued Employment (PACE) program. Under PACE, staff from the Office of Economic Development is co-located at several housing developments to provide employment-related services to residents.

#### 2) Boston

The housing authority recognizes that many residents have numerous psychological and social problems that affect their ability to work toward self-sufficiency. Providing a multitude of services, including employment and training and barrier amelioration assistance, to a diverse and dispersed population, has been a challenge. The housing authority does not have the funds to offer on-site training and social services at every development. Nor does the authority believe that it should be funding or providing services that are available either in the immediate neighborhoods surrounding the developments or accessible by public transportation. The housing authority's efforts to help residents attain self-sufficiency are highly decentralized and guided by three principles. First, residents should use services that are widely available off-site. Second, the housing authority's role is primarily as a referral agent to other service providers. Finally, the tenant organizations should take the lead on addressing the needs of their residents. High level housing authority staff has not coordinated employment-related services with their counterparts at the welfare and labor agencies. However, resident associations and development-level staff have established a relationship with these agencies.

HUD programs are not allowable work activities for TANF clients required to participate in the work program.<sup>16</sup> Cash receipt is limited to 24 months during a continuous 60-month period. The work requirement begins after 60 days of benefit receipt. Work is defined as 20 or more hours per week of unsubsidized employment, subsidized employment, community service, or combining work and community service.

<sup>&</sup>lt;sup>16</sup> Recipients with children who are mandatory full-time school age.

### 3) Los Angeles

The housing authority aims to create communities that focus on economic development and selfsufficiency. Accordingly, public and assisted housing should be transitional, and afford residents the opportunity to work toward accumulating capital and moving into their own homes. The housing authority is creating an infrastructure within individual developments that aids selfsufficiency, including child care centers, computer learning centers, and transportation assistance. The housing authority also has a formal relationship with over 90 non-profit organizations. Not only do these organizations offer training and other supportive services, they seek to help Section 8 households work toward self-sufficiency. The housing authority is also unique in terms of the extent to which individual developments offer training and employment opportunities to residents through both its own construction company, Kumbaya, and residentmanagement corporation businesses, funded as part of meeting Section 3 goals. In addition to operating a number of HUD employment and training programs, the housing authority obtained funding to develop innovative employment-focused programs that are tailored to the needs of its residents. Most recently, it won a competitive Welfare-to-Work grant from the U.S. Department of Labor. Interagency collaboration between the housing authority and the labor department is also strong. The two agencies worked together to create satellite one-stop career centers at a number of housing developments. Although the welfare department and

housing authority have had little interaction in the past, the advent of welfare reform spurred some collaboration between the two agencies, particularly with regard to the Welfare-to-Work and Jobs Plus programs.

Under California welfare law, individual counties have flexibility in determining services and work requirements for TANF clients. In Los Angeles, a client on welfare is eligible for 18 or 24 months of welfare-to-work services through the Greater Avenues to Independence (GAIN) program, depending upon when welfare receipt began. The GAIN process begins with a three-week job club. If the client does not find a job during this time, the welfare case worker and client develop a welfare-to-work plan. Clients must work for 32 hours per week to maintain benefits. Allowable activities include unsubsidized and subsidized employment, job search, community service, vocational training, and jobs skills training or education directed related to employment. Participation in HUD programs can count for the 32-hour work requirement if the client is enrolled in the program at the time of GAIN orientation. Also, after the initial three-week job club, TANF clients can seek employment and training services through the housing authority so long as the welfare agency certifies participation.

#### 4) Milwaukee

The housing authority's philosophy is that work is healthy --- working adults not only feel better about themselves, they provide positive role models for children and others in the community. The employment and training programs reflect this philosophy. Consistent with the state of Wisconsin's work-first approach, the programs focus on helping residents find employment at the earliest possible time; training is secondary. As such, the housing authority aims to provide residents with tools and services to become economically independent. It adapts HUD programs to meet local needs. Staff strongly believes that local labor market conditions and resident needs should drive program content. For example, if the vast majority of potential employees are women with small children, a program that focuses on placing its graduates in construction jobs that have long hours away from home would not make sense. Once the housing authority determines the services that are needed, the programs are shaped accordingly. In addition, the housing authority views on-site service providers as bridges to the community. Currently, it has relationships with over 20 community-based organizations that provide on-site child care, crime prevention, health and nutrition, employment, recreational, and social service programs. Every family development has service providers located on site. The housing authority has a history of collaborating with the County welfare department and the agencies that operate the one-stop job centers. Agency staff is on site at two of the five family developments to provide eligibility determinations for TANF, Food Stamps, Medicaid, and child care.

In Wisconsin, the welfare system leaves little room for education and training. A client who applies for welfare is assessed by a case worker and placed on one of four employment ladder rungs: unsubsidized employment, subsidized employment, community service jobs, or transition jobs. Work begins immediately. There is a two-year time limit for each rung and a 60-month lifetime limit on assistance. HUD programs do not count toward the work requirement.

#### 5) San Antonio

The housing authority is committed to building and maintaining affordable housing and creating safe neighborhoods. More recently, it has begun to encourage self-sufficiency and homeownership among residents of developments and recipients of Section 8. To achieve these ends, the housing authority partners with a variety of organizations to provide housing, education, and employment opportunities for residents. The housing authority has established linkages with CBOs for education and support services, including parenting education and child care. The authority also uses HUD funding streams and its operating budget to craft innovative programs. The Drug Elimination Grant funds resident employment initiatives. HOPE VI funds contribute to the development of one-stop career centers at nine housing projects. The housing authority signed a memorandum of agreement with the local welfare and labor agencies to provide services at these centers, including TANF and Food Stamp eligibility assessments, job search assistance, and job retention services. An operating budget-funded training program prepares residents for clerical and maintenance positions within the housing authority. Under Texas welfare law, the time clock begins when a client enrolls in the welfare-to-work program, Choices. Time limits range from 12 to 36 months, depending upon a client's education and work experience. Adults participating in Choices must engage in work activities for 25 hours per week. The goal is employment at the earliest possible opportunity, so most clients engage in job readiness training or job search. Education and training is permitted only when an individual cannot secure unsubsidized employment. HUD programs do not count toward the TANF work requirement.

## THIS PAGE BLANK

#### CHAPTER III: HOUSING AUTHORITY EMPLOYMENT AND TRAINING SERVICES FOR TANF RESIDENTS OFTEN INVOLVE WELFARE AND LABOR AGENCIES

## **D.** Introduction

This study concerns the degree to which HUD employment and training programs have adapted to the new welfare environment. However, housing authority programs do not operate in a vacuum. Their ability to help residents move from welfare to work depends in large measure on the extent to which they are connected to the broader employment and training systems established by the welfare and labor agencies.

Collaboration among the housing authority, labor department, and welfare department makes sense intuitively. They share common populations, and most housing authority residents are eligible for the labor department's JTPA-funded services (determined by income, number of weeks unemployed, and/or reason for job loss). The issue is not so clear cut, however. Welfare and labor departments often have separate, parallel systems that serve overlapping populations. HUD programs that serve TANF recipients need to find ways to collaborate productively with these other services.

The environment in which housing authority programs operate covered the spectrum from separate welfare and labor employment and training systems to one fully integrated system. In Los Angeles, for example, the welfare and labor departments operated separate employment and training systems. In two sites, Baltimore and Boston, the welfare department contracted with the labor department to provide employment and training services to its clients. In San Antonio, the labor department operates all employment and training programs. Finally, in Milwaukee, the welfare and labor departments are integrated into a single entity.

We noted some strong collaborative efforts among housing authorities and labor and welfare agencies to help welfare recipients attain self-sufficiency in three sites. These generally involved the housing authority working with the labor department, although in some instances the welfare agency also participated. One HUD program, Jobs Plus, required welfare and labor department collaboration.

Finally, we observed more limited collaboration between programs that target Section 8 clients and the welfare and labor departments.

#### E. Collaboration varies across sites

We found varying degrees of collaboration between the housing authority and welfare and/or labor agencies in all five sites. In some cases, collaboration occurs at the administrative level. That is, senior housing authority and welfare or labor agency staff worked together to design or fund programs. In other instances, service delivery-level staff works together at the program or development level.

# 1) Baltimore and Milwaukee had the highest level of collaboration with the welfare and labor agencies

In Baltimore, integration among employment and training programs for TANF recipients begins at the highest level. For example, the housing authority has worked with the labor department, the Office for Employment Development (OED), and the welfare agency, the Department of Social Services (DSS), to design programs for residents, including Step-Up and Jobs Plus. OED

and the housing authority also collaborated to create an employment service program at targeted developments. Staff from both agencies is co-located at selected developments. In addition, OED and DSS provide funds for certain housing authority employment and training initiatives.

"Grant diversion," or the process of using the grants of participants on TANF to offset the cost of a program, is employed for some housing authority programs, and a Memorandum of Agreement between DSS and the housing authority outlines the specific arrangements between the two organizations.

In Milwaukee, housing authority staff described a good relationship with the Milwaukee County Department of Human Services (MCDHS) and the five agencies with contracts to administer the TANF program and the one-stop job centers ("W-2 Agencies"). The housing authority staff and MCDHS have worked together on planning efforts, such as determining the types of services to offer at a HOPE VI site. When welfare reform was implemented, the housing authority served as a site for about 200 community service placements. In addition, both MCDHS and two W-2 agencies now co-locate staff at two developments. MCDHS provides eligibility determinations for food stamps and Medicaid; the W-2 agencies perform initial client assessments, work with clients to create employability plans, and hold employment-related workshops.

# 2) Collaboration between the housing authority and labor agency is strong in Los Angeles

The housing authority and the labor agency, the Community Development Department (CDD), have engaged in a number of joint planning efforts. For example, the housing authority is under contract to CDD to provide satellites to the one-stop career centers at a number of developments. CDD also provides funding for the housing authority's Welfare-to-Work program (described in more detail below). While interaction between the housing authority and the welfare agency, the Department of Public and Social Services (DPSS), was limited in the past, the two entities are beginning to collaborate. DPSS, along with CDD, helped design the Jobs Plus program and is co-locating staff at the two Jobs Plus developments. DPSS also certifies the eligibility of TANF clients who want to participate in the Welfare-to-Work program.

#### 3) San Antonio is expanding its relationship with the welfare and labor agencies

The housing authority and labor agency, Alamo Workforce Development (AWD), have worked together on training initiatives in the past. AWD's vendors, for example, provide training for the housing authority's maintenance/repair training program and Economic Development program (funded by the DEG). The relationship between the housing authority and the local welfare agency, the Texas Department of Human Services (TDHS), however, has consisted primarily of sharing information. Recently, the housing authority, TDHS, and AWD collaborated on a new initiative, the ROAD Centers, which will provide one-stop services at nine housing developments. (The Centers are described in more detail below.)

#### 4) Collaboration in Boston is limited

Senior staff at the housing authority has not engaged in collaborative efforts with their counterparts at the Mayor's Office of Jobs and Community Services or the Private Industry Council, the two entities that administer employment and training services in Boston as well as the three one-stop career centers. Nor does the housing authority coordinate employment and training efforts with the welfare agency, the Department of Transitional Assistance (DTA). Although high-level staff does not work together, there is evidence of coordination at the housing development level. For example, staff members at one of the HOPE VI developments

contact their counterparts at DTA regularly to secure child care vouchers for residents. There is also evidence of collaboration among front-line housing authority and labor staff. FSS coordinators participate in monthly welfare-to-work advisory meetings convened by the Private Industry Council. In the future, the staff from the one-stop career centers will provide on-site assessments of residents at the seven developments targeted by the EDSS grant.

### F. HUD Programs generally do not count toward TANF work requirements

Participation in HUD programs was an allowable work activity for welfare clients in only one site, Baltimore. In a second site, Los Angeles, the welfare office could approve participation only after job search activities proved unsuccessful. The three remaining sites had welfare rules that were heavily work-focused and did not allow residents' participation in HUD program to count toward the TANF work requirements.

In Baltimore, welfare clients are assessed by DSS for education, work history, and need for supportive services, before being referred to other agencies or organizations for training services. DSS will accept participation in all of the housing authority's programs as meeting the work requirement. In addition, the housing authority is negotiating with DSS to become a vendor, meaning it would receive funds from DSS to provide employment and training services for welfare recipients.

The welfare program in Los Angeles, Greater Avenues for Independence (GAIN), begins with a three-week job club. If the client does not find employment during this time, the welfare case worker and the client develop a welfare-to-work plan, which may include job search, work experience, or education and training. Once the plan is in place, TANF recipients can seek employment and training services through other venues, including the housing authority or the city's one-stop career centers. For participation to count toward the TANF work requirement, housing authority program staff or one-stop staff needs to certify to DPSS that residents are, in fact, participating in approved activities.

In the remaining three sites, acceptable work activities for TANF clients are generally restricted to job search. Once employed, even at a low-wage job, TANF clients often earn too much to receive cash welfare payments, although they may still be eligible for supportive services. In Boston, welfare clients who are not exempt from the time limit and required to participate in the work program<sup>17</sup> must work in unsubsidized employment or community service jobs for 20 or more hours per week after 60 days of benefit receipt. These requirements leave little time for education or training. Job placement begins at the DTA office with a brief assessment. Once a case worker determines that a client is not exempt from work, the client must begin a job search. This might involve referrals to one of three one-stop career centers for a more in-depth assessment, job search, and job placement services. A client who cannot find unsubsidized employment can participate in subsidized employment or work experience. In general, housing authority programs are not allowable work activities, although, as described below, DTA will make exceptions on a case-by-case basis.

Milwaukee TANF recipients are required to begin a work activity immediately. A client who applies for welfare visits the job center in her district of the city, where she is assessed by a case worker and placed on one of four employment ladder rungs: unsubsidized employment, subsidized employment, community service jobs, or transition jobs. In addition to determining

<sup>&</sup>lt;sup>17</sup> Clients with children who are mandatory full-time school age must participate in the work program.

TANF eligibility, the job center provides employment services. The housing authority's employment and training programs do not count toward the work requirement.

TANF recipients in San Antonio are also required to engage in work activities for 25 hours per week. The goal is employment at the earliest possible opportunity, so most clients engage in job readiness training or job search. Education and training are permitted only when an individual cannot secure unsubsidized employment. Given the strong work focus of welfare, the housing authority's employment and training programs do not count toward the TANF work requirement.

# G. Primary focus of collaboration on employment and training services for TANF recipients is often outside of discrete HUD programs

In three sites, the major housing authority efforts to help TANF recipients with employment and training needs fall outside of specific HUD programs. All three involve active collaboration with the local departments of labor and, in one instance, the welfare department.

#### 1) Los Angeles Welfare-to-Work and One-Stop Center "Satellites"

In Los Angeles, the housing authority's most significant program for residents receiving TANF is the Welfare-to-Work program, funded by a competitive Department of Labor Welfare-to-Work grant. This program involves collaboration with CDD and, more recently, DPSS. The program aims to place 1,000 residents in permanent, unsubsidized employment by December 31, 2000. The program includes assessment, case management, job placement services, and, for those who are not job-ready, work experience placements. CDD provides funding and space for Welfare-to-Work staff. While the primary funding source is a competitive Department of Labor grant, the Private Industry Council, which, along with CDD, administers employment and training services, provides funding to serve an additional 300 individuals.

For TANF recipients to receive employment and training services through the welfare-to-work program, DPSS must certify their eligibility. This can happen in one of two ways: DPSS can directly refer TANF clients to the program for services, or the housing authority can recruit eligible clients and then seek DPSS certification ("reverse referrals"). While the number of direct referrals from DPSS has steadily increased, approximately 85 percent of clients are still recruited via reverse referrals. DPSS co-located two staff members to the Welfare-to-Work program once a week to help process new clients and resolve any problems.

Another major employment and training initiative in Los Angeles is the one-stop "satellites." The housing authority, under contract with CDD, established on-site employment and training service centers at five of its housing developments and the Family Investment Center. In addition, two new satellites are opening at the Jobs Plus developments. The housing authority receives \$3.6 million annually from CDD for the satellites.

Generally, one-stop satellite employment and training services include employment placement, referrals for education and training (on-the-job and classroom) to CDD vendors, referrals to social service providers, career resource centers (with job listings, phone banks, Internet access), workshops, and job clubs. While in training, participants are eligible for assistance with child care and transportation.

#### 2) Baltimore PACE Program

In Baltimore, the housing authority and OED created the People Accessing Continued Employment (PACE) program at four developments that house HUD programs: two FIC sites, a HOPE VI site, and the Jobs Plus development. PACE provides pre-placement, post-placement, and job-replacement services, as well as support services and job clubs for the residents. OED staffs and operates the program.

PACE focuses on the entire family, as opposed to focusing solely on the employment and training needs of the leaseholder. Along these lines, PACE identifies and deals with barriers to employment up-front, before focusing on job search. PACE also focuses on training for a career. Staff inquires about work histories and develops employment approaches accordingly. The PACE Job Club is designed as a support for people who are already working.

### 3) San Antonio ROAD Centers

In San Antonio, the Resident Opportunities for Achievement and Development (ROAD) Centers involve a collaboration among the housing authority, AWD, and TDHS. The collaboration among the three agencies began after the passage of the state's welfare reform law. The housing authority is providing space for the Centers, while the other partners are contributing staff. Each ROAD Center will provide on-site case management, and a range of job search tools including Internet access, job descriptions and applications, assistance to complete applications, resume writing assistance, and access to basic skills training. The first ROAD Center opened in summer 1999; the remaining eight will be on line by September 2000. Each agency will provide the following services:

- □ The housing authority will contribute space at nine developments for the ROAD Centers. In addition, FSS coordinators will be located on-site to provide referrals to educational training, employment, and supportive services.
- TDHS will provide on-site staff to make eligibility determinations for TANF, Food Stamps, and Medicaid. In addition, a Texas Works Community Service Advisor will coordinate with other staff to reach residents most in need of services. (Three Advisors will serve nine ROAD Centers.)
- AWD, through its contractor, SER-Jobs for Progress, will provide an Assessment Specialist and Case Manager at each site. The Case Manager will assist clients with entry into the labor market, including job readiness training, job placement, post-employment services, and educational advancement.

#### 4) Milwaukee Resident Employment Coordinators

The Milwaukee housing authority uses EDSS and HOPE VI grants to fund Resident Employment Coordinators who provide job search assistance to residents, including those on TANF, in the five family developments. The coordinators work in tandem with staff from two of the five W-2 agencies.

The resident employment coordinator conducts an in-depth assessment that focuses on the job interests of the resident, assists the resident with his or her job search, and makes referrals for services that address employability skills and barriers to employment such as child care and transportation.

In addition, county welfare department staff and W-2 agency staff are on-site at two of the five developments. County staff members assist residents with issues relating to child care, Food Stamps, and Medicaid. Staff from two W-2 agencies help residents find employment. One W-2 agency, MAXIMUS, contracted with the housing authority to have the resident employment coordinators provide job search assistance to its entire TANF caseload (including recipients who do not reside in public housing). Every week MAXIMUS staff forwards a list of clients who are

employment ready and facing expiring benefits to the resident employment coordinator. The housing authority receives a payment from MAXIMUS for each placement that is made.

# H. Jobs Plus provides a strong focus on collaboration with welfare and labor agencies

Jobs Plus was the single HUD employment and training program that collaborated with both the welfare and labor departments in all sites that we observed. According to HUD program regulations, the labor and welfare departments are mandatory partners in the Jobs Plus collaborative, which is responsible for designing, funding, and operating the program. We observed Jobs Plus in Los Angeles and Baltimore.

In Los Angeles, Jobs Plus was a vehicle to begin involving the welfare agency, DPSS, in the housing authority's programs. Prior to this initiative, housing authority collaboration was limited to CDD, the labor agency. The housing authority provides space for CDD and DPSS at both Jobs Plus developments to provide on-site services. CDD opened a one-stop satellite at each location. DPSS staff is on site at each development once per month and plans to increase site visits to once per week. In addition, all Jobs Plus participants are assigned to one DPSS case manager.

In Baltimore, DSS and OED helped design the program and provided funding. DSS worked with the housing authority to structure program activities that would count toward the work requirement. DSS also refers its clients residing in the Jobs Plus development to the program for employment and training services if they are not already working with another vendor. In addition, DSS is training Jobs Plus staff to process child care vouchers for its participants. The Jobs Plus development also has the PACE program on site, which is staffed by OED.

# I. Less evidence of coordination and collaboration for programs that target Section 8 residents

The programs that target Section 8 residents—FSS, Neighborhood Networks, and MTO generally do not involve active partnerships with labor or welfare departments. For the most part, these programs are not work-focused; few meet TANF work requirements. The Family Self-Sufficiency programs in Boston, Los Angeles, and Milwaukee focused exclusively on Section 8 residents. In each instance, the program did not count toward the TANF work requirement. Only in Baltimore, which targeted both development-based residents and holders of Section 8 vouchers, did FSS count toward the work requirement.<sup>18</sup> Generally speaking, FSS staff did not report regular contact with their counterparts in the welfare and labor departments. This will change in San Antonio, when the ROAD Centers are fully operational. FSS staff will be housed on site along with staff from the welfare and labor departments. Boston FSS staff also has limited contact with the labor department. They participate in the monthly welfare-to-work advisory meetings hosted by the Private Industry Council. The Neighborhood Networks program offers computer facilities and training at developments receiving Section 8 project-based assistance. We observed the program in three sites. None formally collaborated with their respective welfare and labor departments to plan or fund

<sup>&</sup>lt;sup>18</sup> As indicated earlier, the housing authority and welfare department have an agreement to count all housing authority programs toward the work requirement.

services, however, two sites had more informal contact. In Baltimore, for instance, the Neighborhood Networks center shares space and works closely with the Urban League, a welfare department vendor. Program participants interested in employment assistance can easily access services through the Urban League. In Boston, as noted earlier, training activities do not generally count toward the 20-hour per week work requirement. However, TANF recipients can obtain waivers to participate in training programs. The director of the Neighborhood Networks center facilitates this process by intervening with welfare caseworkers on behalf of TANF recipients who wish to participate in the computer training program. However, the Boston staff believes that the welfare case workers like the Neighborhood Networks program because there is a clear link between training and job placement. In addition, the Center has provided community service placements for TANF recipients who could not find unsubsidized employment. Finally, the director talks regularly with welfare office staff regarding participants' child care problems. The Los Angeles site had no contact with either the labor or welfare departments. Finally, Moving to Opportunity provides Section 8 vouchers to families in targeted developments to enable them to move to low-poverty neighborhoods. We observed the program in Boston and Los Angeles. Neither site collaborated with the local welfare and labor

developments to enable them to move to low-poverty neighborhoods. We observed the program in Boston and Los Angeles. Neither site collaborated with the local welfare and labor departments. In each site, the housing authority contracted with a non-profit organization to provide housing search assistance. Following the move, the non-profit group offered supportive services including counseling, job development workshops (e.g. resume development, Internet searches), transportation assistance, and referrals for social services and/or education and training. In both sites, job-related assistance often entailed referrals to the local one-stop career center. However, participation in the post-move services is not mandatory. Staff in both sites noted that most of their time was spent on housing-related assistance, and that once accommodations were settled, the level of services requested decreased considerably. Staff did not find this surprising, remarking that MTO is a housing program and not a work program.

### CHAPTER IV: HUD PROGRAM ALIGNMENT WITH FINDINGS FROM EVALUATIONS OF EMPLOYMENT AND TRAINING PROGRAMS

## J. Introduction

As indicated in Chapter II, HUD's employment and training programs use a variety of strategies to address the needs of public and assisted housing residents. Although HUD's efforts at helping families in public and assisted housing have a long history, many of its programs have not been examined in light of recent research on the effectiveness of various welfare-to-work strategies. The Success in the New Welfare Environment project involved assessing HUD's efforts to help residents transition from welfare to work. As with any assessment, one must establish a benchmark—a basis for comparison. Simply asking if HUD's programs are "effective" misses the larger point. Programs are complex and comprise multiple components. Some individual program elements might be more effective at helping welfare clients obtain work than other ones.

Thus, we began this project by reviewing the literature on programs that have attempted to help welfare recipients find jobs, remain in jobs, and advance in the labor market. Based on this review, we identified a number of program components that seemed to be associated with positive outcomes—exits from welfare, increased rates of employment, or increased earnings. These elements provided the basis for the framework that was used to identify and assess HUD's employment and training programs.

This chapter reviews how HUD's employment and training programs compare to the elements of effective programs that were identified in the literature. It begins with a brief description of the assessment framework. The remaining sections of this chapter focus on each component of the framework. The outline for each of these sections is as follows:

- □ A summary of the research findings.
- □ A description of the degree to which individual HUD programs incorporated the element and, when applicable, housing authority initiatives we observed outside the scope of the HUD programs under study.

#### 1) Assessment Framework

There is a significant body of evaluation literature on employment and training services for welfare recipients. (See Appendix A, the Literature Review, for a more thorough discussion of the research.) Most of the literature that is available focuses on pre-employment services or identifying barriers to employment. Because of this focus, the studies provide information on whether participants find jobs, how much they earn, welfare receipt (number of months and/or amount received), and, occasionally, whether they keep those jobs for some period of time after treatment. They do not necessarily provide information on whether these programs promote self-sufficiency.

Moreover, the bulk of the research was conducted under the rules of the AFDC program, not under the time-limited environment of TANF. It is likely that the time limits will affect participants' behavior (e.g. they may be less likely to leave an initial job). It is equally likely that states will be serving a more disadvantaged population than they have in the past, as those who might previously have been exempted from participation in employment and training programs will now have to participate.

With these caveats in mind, the evaluation research suggested that some employment and training programs have placed participants in jobs, increased earnings, and reduced welfare benefits. However, even the most successful programs have had difficulty sustaining these positive impacts over time. These evaluations also indicated that there are a number of program components and strategies that are associated with positive employment outcomes. The elements can be grouped into four broad areas, or service clusters:

- **Employment services** might have an employment or education and training focus. Specific components include:
  - Job search
  - Work experience
  - Subsidized employment
  - On-the-job training
  - Vocational training
  - Basic skills training
  - Post-secondary education
  - Assessment
  - Case management
- **Barrier amelioration services**, which focus on:
  - Environmental barriers (child care and transportation)
  - Personal barriers (substance abuse, mental health problems, lack of social support, and domestic violence)
- **Cash and in-kind supports**, which augment earned income and include:
  - Earned income disregards
  - The Earned Income Tax Credit (EITC)
  - Food Stamps and other nutrition programs
  - Transitional Medicaid
- **Transitional supports**, which address a range of workplace issues and include:
  - Life skills courses
  - Support groups

**Table 4.1** describes each component and the findings from the evaluation literature.

Table 4.1	Assessment	Framework
-----------	------------	-----------

	COMPONENT	EVALUATION FINDINGS		
ent Services	Employment- vs. Education and Training- focused Strategies utilize different means to help people work toward self-sufficiency. Employment- focused programs may still incorporate <i>short-term</i> (3 months or less) education and training. Job Search may be characterized by low intensity individual search or high intensity search which may involve working with a job assistant who provides training in resume writing, interviewing skills, and/or promising job search tactics.		These strategies were tested through many of the <i>JOBS</i> programs. Findings indicate that employment- focused programs can moderately increase employment and earnings and decrease welfare receipt and payments. Education and training-focused programs generally have produced inconsistent results. There is some evidence that impacts may increase over time. This is a costly, time consuming intervention. Mixed strategy approaches, such as that used in <i>Oregon JOBS</i> , may hold the most promise. Low to moderate intensity job search (including unassisted individual job search or job club followed by independent job search) as conducted in the <i>Arkansas WORK</i> program, did have positive, significant two- year impacts on employment and earnings. However, the impacts disappeared by the end of the fifth year. The impacts of this program were small relative to more comprehensive programs that offered more services in addition to job search. Gueron and Pauly (1991) found that job search assistance does generally increase employment but that it does not affect wages. They also conclude that job search assistance <i>alone</i> does not substantially reduce welfare dependence. Job search, as used in <i>California GAIN</i> , <i>Project Match</i> , and others, is often deemed an essential piece of a "successful" program.	
Employme	Work Experience / Subsidized Employment both aim to provide the worker with employment skills. Work Experience (or workfare) entails working in "exchange" for benefits and is generally public sector employment. Subsidized employment pays the worker wages and often takes place in the private sector.		Successful program.Workfare (low intensity work experience) which was utilized in the WIN and JOBS programs generally lasted1-3 months and was found to have no consistent impacts on employment, earnings, or welfare receipt.In the Baltimore Options program, it was determined that short-term work experience alone had no impactson employment or welfare receipt.However, the program did have positive impacts overall.The National Supported Work Demonstration provided higher intensity work experience in the form ofsubsidized work and had significant, positive impacts on earnings for many years.It also increasedemployment and reduced welfare receipt.The Home Health Aide demonstration, which involved paid supported work experience, also had significantimpacts.It increased earnings and employment rates while reducing welfare receipt at four of seven sites.	
	<b>On-the-Job Training (OJT)</b> is generally employer provided, subsidized training in the workplace. It may be provided as a formal training program prior to employment or in conjunction with work.		On-the-job training (OJT) has had consistent positive impacts on employment and earnings of welfare recipients, namely the "hardest-to-serve." Programs which utilized OJT, such as <i>Maine TOPS</i> , <i>Baltimore Options</i> , <i>California GAIN</i> , and <i>JTPA</i> found earnings gains, some of which were sustained beyond the short-term (two years). Few, however, had significant impacts on employment.	
	Vocational Training aims to train a potential worker for a specific job or trade. It often entails classroom training and may be combined with on- the-job training.		The Center for Employment Training (CET) site in the <i>Minority Female Single Parent Demonstration (MFSPD)</i> focused on occupational training and education. It had promising initial earnings results. <i>Child Comprehensive Development Program (CCDP), Oregon</i> and other <i>JOBS</i> programs, <i>California GAIN,</i> and <i>Maine TOPS</i> all utilized vocational training and had positive impacts on employment or earnings. The <i>New Chance</i> program, however, utilized vocational training and had no significant impacts on employment or earnings.	

## Table 4.1: Assessment Framework (continued)

	COMPONENT	EVALUATION FINDINGS	
	Basic Skills is considered to be "general" education and is not targeted to a specific job,		Programs that focus on basic skills tend to have few, if any, positive impacts on employment, earnings, or reduced welfare receipt.
es	industry, or trade. It may involve remedial education, English as a Second Language (ESL), Adult Basic Education (ABE) and General		Some more comprehensive programs which include basic education, <i>California GAIN</i> , <i>Baltimore Options</i> , and the <i>JOBS</i> programs, for example, have had positive results. Yet, it is not evident that they are attributable to the presence of education services.
Ö	Educational Development (GED) training.		The <i>Even Start Family Literacy Program,</i> which focused on basic education, had no positive impacts on the above measures of self-sufficiency.
Servi			While some programs such as <i>Oregon JOBS</i> , <i>New Chance</i> and The <i>Even Start Family Literacy Program</i> did increase GED attainment, the research shows that when years of schooling and work experience are controlled for, GED certification has no positive impact on earnings or employment.
ent			Programs such as <i>California GAIN</i> and <i>JOBS</i> do, however, tend to have stronger positive impacts for subgroups of people determined not to need basic education.
<b>B</b>	Post-secondary Education generally involves		There is little evidence supporting the impacts of post-secondary education as very few programs offer it.
ЭЩ	college level courses/education after receiving a high school diploma or GED certificate.		The <i>Minnesota Family Investment Program</i> offered post-secondary education; however, its positive impacts are generally attributed to the financial incentives.
0			The <i>Quantum Opportunity Program</i> made post-secondary education a goal. The program led to an increase in skills, graduation from high school, and continuation in post-secondary education.
mploym	<u>Assessment</u> generally entails an interview with a case worker to determine general characteristics of participant. Higher intensity assessment may be		There is little experimental data to support the notion that assessment directly impacts increased employment and earnings. However, it is assumed to be a valuable tool used to gauge employment skills and interests, reveal barriers, and target services.
ш	repeated and include barrier screening and skills testing.		As with the <i>JOBS</i> demonstrations, assessment intensity varies depending on the program's focus. As a result, impacts are not solely attributable to assessment type.
	5		Two targeting approaches are currently in the formulation and evaluation stages. They include (1) The Worker Profiling and Reemployment Services system and (2) a checklist-type targeting tool developed by Mathematica Policy Research.
	<u>Case Management</u> can range from low to high intensity. Low intensity case management is		Many programs utilize case managers to help welfare recipients get the services they need. Evaluations of the <i>Postemployment Services Demonstration</i> and the <i>Family Transition Program</i> found that
	characterized by occasional or rare contact with a case worker. Medium to high intensity case management involves more frequent contact (monthly or weekly) and may involve home visits,		welfare recipients will use case management services, at least in some circumstances. Thus far, there is no evaluation evidence that links case management directly with employment outcomes.
	follow-up with employers, service referrals and follow-up with service providers.		

## Table 4.1: Assessment Framework (continued)

BARRIER <sup>19</sup>	EVALUATION FINDINGS	
Child Care	The current supply of infant care is insufficient.	
L L	The supply of care during non-standard hours is also insufficient.	
	The type of provider is related to duration of work; women with children in center-based care or non-relative care	
	worked for longer periods of time than those who used relative care.	
	Evaluation literature concerning the implications on employment, earnings, and self-sufficiency as related to child	
Transportation Social Support	care is sparse.	
Transportation	A spatial mismatch exists between jobs and residents.	
	Transportation routes or hours are often insufficient.	
	Programs are being tested (Bridges to Work and Joblinks), but thus far no evaluation results are available.	
Social Support	A lack of support networks can lead to depression and job loss according to case studies.	
Ξ	Providing a network is an aspect of many programs, such as New Chance and the Quantum Opportunity	
	Program, and while such provisions appear to be common sense, they are difficult to evaluate.	
<b>O</b> Domestic Violence	Prevalence estimates range from 15% to 32% of current welfare recipients.	
	□ States and programs are experimenting with methods to screen abused women and refer them to care; thus far,	
	there is no evaluation evidence.	
Mental Health	Welfare recipients are more prone to depression, anxiety, and low self-esteem than the general population.	
	Depression can be effectively treated.	
	□ States are experimenting with screening mechanisms and referrals, but no evaluation evidence of what "works"	
al	is available.	
	Employee Assistance Programs for those who work in medium or large firms are a possibility.	
Substance Abuse	Welfare recipients are more likely to have a substance abuse problem than the general population.	
	□ States, particularly Oregon and Utah, are experimenting with the use of screening mechanisms and referrals.	
	Employee Assistance Programs for those who work in medium or large firms are a possibility.	

<sup>&</sup>lt;sup>19</sup> Many programs, including EDSS, TOP, Step-up, FIC and Youthbuild do address "employment barriers," but do not specify the provision of services for any specific barriers. Local programs could use this funding for any of these services.

## Table 4.1: Assessment Framework (continued)

	COMPONENT	EVALUATION FINDINGS
ional rt	Life Skills Training can involve an array of services which may include, but is not limited to, training or assistance with parenting, budgeting, conflict resolution, family planning, and soft skills (e.g. job etiquette, timelinesset cetera)	<ul> <li>Such training is incorporated into many programs because it is perceived to be necessary.</li> <li>However, little empirical evidence is available concerning the benefits of, or the need for, life skills training.</li> </ul>
<b>Transiti</b> Support	Support Groups and Mentoring involves group discussion about personal and work related issues and one-on-one guidance, respectively. Both components are largely peer driven and tend to rely on trust as a major factor of success.	<ul> <li>Case studies have found that women utilize support groups, and that having a regular time to talk through problems can help prevent emergencies. However, these interventions have not been rigorously tested.</li> <li>When provided by experienced programs with well-developed standards, one-on-one mentoring has been found to help adolescents avoid risky behaviors (e.g. drug and alcohol use).</li> <li>Mentoring programs that pair successful adults with those entering the workforce are largely untested.</li> </ul>

These components became the basis for the framework used to assess the programs and activities used by housing authorities and property managers to help residents achieve self-sufficiency in the new welfare environment.



#### Figure 1: Systems Approach

The framework is depicted in Figure 1.<sup>20</sup> It illustrates the fact that individual program components are just one part of the assessment. Beyond the success of specific program components, there is a larger issue facing housing authorities as they seek to move residents from welfare to work. Many residents will need a wide variety of services and supports to find employment and work toward self-sufficiency. At any given point in time, a client may need assistance in accessing one or more of the identified service clusters. Furthermore, the breadth of services and public systems involved complicates the provision of services and increases the importance of creating linkages with other providers. Services and supports may be required from the welfare/human services system, the employment and training system, the health care system (including mental health and substance abuse services), the transportation system, the education system (secondary, vocational, and post-secondary), as well as the housing system. The remainder of this chapter focuses on individual program components.

#### K. Program Philosophy

#### 1) Summary of Research

Much of the research regarding employment services focuses on the preferred strategy for providing those services – an employment-focused strategy versus an education and training-focused approach.

**Employment-focused programs** range from minimal, job search-only programs to mixedstrategy programs, in which other services, such as short-term education and training, may be provided. In keeping with the philosophy of these programs, the focus of each element, including any education and training that may be provided, is on getting clients into the workforce as quickly as possible.

**Education and training-focused programs**, on the other hand, are designed to improve present and future employability by improving basic skills and providing training and education that increase employment skills. Participants typically undergo a thorough up-front assessment to determine service needs and create an employment development plan. They are then assigned to various service components, such as basic skills training, vocational training, or higher education, based on their individual needs and career goals. Clients are encouraged to be selective and choose a job that fits with their long-term goals.

<sup>&</sup>lt;sup>20</sup> The Assessment Framework is drawn from Fishman, M., Barnow, B., Gardiner, K., Murphy, B. & S. Laud (1999).

Generally, evaluations of welfare-to-work programs suggest that employment-focused programs can be effective at connecting TANF recipients to the work force quickly. For those who are not likely to leave welfare on their own, studies of employment-focused programs indicate that they can moderately increase employment and earnings and reduce welfare payments. This is especially true of participants in "mixed strategy" programs, which combine job search with other services such as short-term education or intensive case management. Evaluations of job search-only programs tended to show smaller impacts. Research on education and trainingfocused programs has not found consistent results. While there is some evidence that impacts for these program participants may increase in the long run, it was found that after two years the earnings gains were still smaller than those achieved by employment-focused programs. Under both scenarios, employment and earnings gains are higher for those with a high school diploma or GED. Also, earnings impacts are often attributable to higher rates of employment, not to better-paying jobs.

Program Philosophy	HUD Program
Employment-focused	EDSS
	Jobs Plus
	Section 3
Education and training-focused with strong	Step-Up
work component	YAP
	Youthbuild
Education and training-focused	FSS
	Neighborhood Networks

#### 2) Summary of Housing Authority Activities

Consistent with the literature, few HUD programs we reviewed focused solely on education and training.

Three programs were employment-focused: Section 3, Jobs Plus, and EDSS. While none excluded the option of training, the overarching focus of each program was work. The EDSS program, for example, sought to help families achieve self-sufficiency by providing grants to housing authorities for employment-related activities, including job training, job search assistance, job development and placement, and continued follow-up assistance. Similarly, Jobs Plus aimed to create housing developments where the majority of working-aged adults are employed.<sup>21</sup> As such, job services were the central focus, although social and economic supports were also available. Finally, Section 3 mandated that housing authorities use the funds they receive from HUD to generate jobs for residents in low-income communities. The program focuses on connecting residents to employment. The method varied by site. In Los Angeles, for example, the housing authority developed and administers a public construction company, Kumbaya. The two other Section 3 programs, in Baltimore and Boston, referred residents to housing authority contractors when jobs became available.<sup>22</sup>

Three programs, Step-Up, YAP, and Youthbuild, could be classified as education- and trainingfocused with a strong work element. All three provided a combination of hands-on construction

<sup>&</sup>lt;sup>21</sup> Both Jobs Plus programs observed were funded in part with EDSS grants.

 $<sup>^{22}</sup>$  About one-third of Kumbaya's new hires are Section 3 residents. In Boston, 42 percent of new hires at a HOPE VI site under construction are residents.

and/or maintenance experience and classroom work. The programs also helped graduates find construction-related jobs or further apprenticeships in well-paying trades.

Only two programs, FSS and Neighborhood Networks, could be classified as education and training-focused initiatives. FSS helped residents attain self-sufficiency through referrals to education, training, and supportive services. Neighborhood Networks aimed to increase residents' employment opportunities by offering computer access to housing residents. Computer center programs may include computer training, Internet access, job readiness support, GED certification, and adult education classes.

The remaining HUD programs were not easily classified. These include FIC and HOPE VI, both of which provided space to housing authority and non-housing authority service providers for services ranging from basic education to job search. The TOP grant enabled tenant organizations to develop programs to encourage resident self-sufficiency; the degree to which services were employment- or education-focused depended upon the tenant organization. Finally, MTO focused on moving residents to low-poverty neighborhoods; employment and training were secondary activities.

## L. Job Search

## 1) Summary of Research

Traditionally, job search is a pre-employment or re-employment activity and is conducted either individually or as part of a job club or other group job search program. While the primary goal of job search is to get participants employed, job search and job search training can also be used to help participants find better jobs with higher wages.

Job search programs, offered alone or in combination with other services, have consistently increased employment rates. However, while job search is necessary to find any job, past research indicates that job search-only programs do not seem to help participants find jobs that will allow them to become self-sufficient. Programs that offer only job search have rarely led to significant impacts on wages or a reduction in welfare receipt.

Job Search	HUD Program
Primary program component	EDSS
	Jobs Plus
One of many components	FIC
	HOPE VI
	Neighborhood Networks
	Step-Up
	Youthbuild
	YAP
Minimal component	FSS
	МТО

## 2) Summary of Housing Authority Activities

Ten HUD programs offered some form of job search. Consistent with the literature, none focused exclusively on job search. However, we observed a large degree of variation in the role job search played in each individual program.

Job search was the primary component in two programs: EDSS and Jobs Plus. This was true of all sites, although the job search process differed by program and site. Both EDSS sites used or planned to use the grant to hire employment coordinators or counselors. In Milwaukee, for example, the two resident employment coordinators help residents conduct self-directed job searches by assisting with application and resume development, protocol for contacting employers, and mock interviews. Coordinators also develop job leads and will make referrals. In Jobs Plus, both sites—Los Angeles and Baltimore—have on-site job search services including job developers and resource rooms for self-paced job search activities.

Job search was one of many components for FIC, HOPE VI, Neighborhood Networks, Step-Up, Youthbuild, and YAP. In FIC and HOPE VI, space was generally available for non-housing authority staff to offer services. Both FICs in Baltimore and the HOPE VI development, for example, had PACE staff on site. The HOPE VI development in Milwaukee<sup>23</sup> provided space to a W-2 agency. In both FIC and HOPE VI sites, services ranged from resume development to posting job listings. The San Antonio housing authority plans to offer a range of job search tools at its ROAD Centers, including internet access, job descriptions and applications, and staff assistance with resumes and applications.

The construction-related programs—Step-Up, Youthbuild, and YAP—aimed to help participants find better paying jobs by providing them with skills training and assisting them in their job search. In some cases, program staff also assisted participants who were not interested in construction work to find jobs outside of the field. In Neighborhood Networks, efforts to help participants find jobs ranged from an integral part of the training program in one of the three sites to non-existent in another site.

Finally, job search was a minimal program component in FSS and MTO. Generally, FSS coordinators in all five sites refer clients who are seeking employment to the city's employment and training providers, such as the labor department one-stop career center. Likewise, MTO staff in both sites referred clients to the local one-stop career centers for job search services. In addition to the HUD programs under review, the housing authority in Los Angeles offers job search services through its own initiatives. The Welfare-to-Work program provides the housing authority's most intensive job search activities. Housing authority staff work extensively with clients to place them into jobs, meeting as often as one to three times per week. There is a job developer on staff to establish partnerships with private employers, develop internship opportunities and on-the-job training programs, and make job referrals. In addition to the Welfare-to-Work program, the housing authority is under contract from the city's labor department to operate "satellite" one-stop career centers at five developments. These satellites offer employment placement services, career resource rooms, and job clubs.

## M. Work Experience, Subsidized Employment, and On-the-job Training

## 1) Summary of Research

Work experience, subsidized employment, and on-the-job training provide opportunities to gain practical experience in the labor market.

**Work experience**, also known as community work experience, workfare, or community service employment, provides a way for participants who have not been able to find a job to meet TANF

 $<sup>^{23}</sup>$  W-2 agency staff is on-site at one HOPE VI development and will be within walking distance of a second (yet to be completed) development.

work participation requirements. Under these programs, participants are generally required to work in exchange for welfare benefits. The work is often in the public sector.

Under **subsidized employment**, participants are paid wages that are funded from diverted welfare payments. Subsidized jobs can help participants learn skills that might lead to eventual employment. These positions are usually with private sector employers. Subsidized employment and supported work, unlike workfare or work experience, tend to provide a more intensive intervention and include support services in conjunction with the employment experience.

**On-the-job training (OJT)** is employer-provided training in the workplace. Unlike subsidized employment, it is generally available to both welfare and non-welfare participants who are working in unsubsidized jobs that they usually retain after training is completed. OJT is funded through employment and training funds (not through diverted welfare payments) or by employers.

Studies of programs that offer work experience alone have found no consistent impacts on employment, earnings, or welfare receipt. Combining work experience with other training activities, such as on-the-job training, may hold promise. Evaluations of subsidized employment and OJT programs have found significant, positive impacts on employment and earnings. Subsidized employment is also associated with decreases in welfare receipt.

Component	HUD Program
Work experience	FIC
	HOPE VI
	Jobs Plus
Subsidized employment	Step-Up
On-the-job training	Section 3
	YAP

## 2) Summary of Housing Authority Activities

Work experience, subsidized employment and OJT were among the least common program components we observed, even though in three sites, Baltimore, Boston, and Milwaukee, work experience and subsidized employment were allowable TANF activities. When offered, they were generally components of a more comprehensive intervention, such as HOPE VI or an apprenticeship program.

While work experience was not an established component of any program across all sites, three programs--Jobs Plus, HOPE VI, and FIC --included a work experience component in some sites. The FIC program in Milwaukee, for example, had a work experience program that targeted youth. An on-site, non-profit organization provided job readiness skills and job experience, with the expectation that after participants gain work experience they will be able to find jobs independently in the private sector. Work experience was also a component of the Baltimore HOPE VI program. Unemployed TANF recipients could participate in the Community Service Program and train to perform entry-level tasks until they were placed in job-training programs or unsubsidized jobs.<sup>24</sup> The Jobs Plus program in Baltimore also incorporated work experience for clients who were not job-ready.

<sup>&</sup>lt;sup>24</sup> The Community Service Program is operated on site by the welfare department

Only one HUD program offered subsidized employment. The Baltimore Step-Up program utilized funds diverted from welfare grants to pay participant wages.<sup>25</sup> About one-third of Step-Up participants were on public assistance at the time they enrolled in the program, and thus were eligible for subsidized employment.

Finally, we observed on-the-job training provided under the auspices of YAP or Section 3. For example, in all four YAP sites, participants were paid wages during their training. In two of the four sites, Baltimore and Boston, graduates were guaranteed employment with their respective housing authorities. The remaining sites made concerted efforts to place graduates in construction jobs. In two of the three Section 3 sites we observed, Baltimore and Los Angeles, residents often received on-the-job training through the program. In Los Angeles, about 40 percent of the workers at the housing authority's construction company, Kumbaya, were resident apprentices who were paid an hourly wage and received training from the field supervisors. In Baltimore, the housing authority occasionally asked their contractors to provide training to residents hired through Section 3.

Although work experience, subsidized employment, and on-the-job training were not common components of HUD programs under review, we noted examples of these elements among housing authorities' own employment and training initiatives.

For example, Baltimore offers subsidized employment for programs in addition to Step-Up. The housing authority has a Memorandum of Agreement with the welfare agency to use diverted welfare grants to pay for up to 10 months of training. Programs financed through diversion included grounds keeping, pesticide and lawn care training programs.

The Los Angeles housing authority provided work experience and on-the-job training through its Welfare-to-Work program. Residents who were not ready for unsubsidized employment could participate in the welfare-to-work program's work experience jobs. Participants worked for 20 hours per week at the minimum wage, which was funded through the Welfare-to-Work grant. Finally, in Boston, individual housing developments offered work experience placements. We observed one development that places up to 15 residents at a time in work experience jobs, either in the on-site thrift shop or assisting with maintenance work. Other developments offer residents work experience positions involving office tasks such as filing or answering the telephone.

## **N. Vocational Training**

## 1) Summary of Research

Vocational training programs, which train participants for *specific* careers, provide another avenue for welfare recipients to obtain skills that may lead to higher quality jobs. Vocational training programs differ from on-the-job training in that they are not followed by a guaranteed job, although many programs have a job placement component. Participants generally complete some combination of education and on-the-job training to gain the requisite skills for their chosen career. Job placement following training may or may not be part of the program. For the most part, because welfare recipients tend to have low skill levels, vocational training has focused on careers that do not require high skill levels.

Research indicates that some programs that have focused on vocational training have had modest success in helping participants obtain higher wages or work more hours than they may have in the absence of the training.

<sup>&</sup>lt;sup>25</sup> A maximum amount of \$5,000 per grantee can be diverted over 10 months.

The literature also suggests that ease with which program participants find jobs is a function of the demand for the specific vocational skills. Vocational training that is tailored to fit the local labor market will likely result in better employment outcomes.

Vocational Training	HUD Program
Long-term	Step-Up
	Youthbuild
Short-term	FIC
	HOPE VI
	Jobs Plus
	Neighborhood Networks
	Section 3
	Step-Up

### 2) Summary of Housing Authority Activities

Two programs, Youthbuild and Step-Up, offered long-term vocational training. Both aimed to prepare clients for well-paying jobs in the construction trades. The Step-Up program in Baltimore entailed a two-month pre-training period, which focused on job readiness (e.g., workplace behaviors and budgeting) and construction-related issues (e.g., blueprint reading, safety, and tool identification), followed by one year of on-site training with skilled journeymen. During the apprenticeship portion of the program, each participant was guaranteed rotations in four trades: carpentry, plumbing, electrical, and masonry. Apprentices were paid \$9.00 per hour. Similarly, all five Youthbuild programs generally provided one year of hands-on basic carpentry and construction skills training (including flooring, roofing, dry walling, framing, and finishing) and classroom-based education. Participants also received wages during the training period, ranging from a stipend in some sites to \$6.00 per hour in others.

Six programs offered short-term training. Neighborhood Networks, for example, provided training in computer skills. In two of the three sites we observed, classroom training lasted for six weeks and encompassed computer "literacy" (e.g., how to use the mouse, double clicking, and cutting and pasting), word processing, spreadsheets, and graphics. The third program we visited was self-paced and lasted from two weeks to six months, depending on the client's skill level and previous experience. In some housing authorities, the FIC and HOPE VI programs offered on-site training in fields such as computer skills, child care, home health care, or security.

Other programs worked with local colleges to design short-term training initiatives to meet community needs. The Baltimore Jobs Plus program, for instance, worked with the Maryland Center for Arts and Technology to assemble a 16-week training course at the behest of a local employer. Participants were trained for customer services and finance positions. Another program targeted 17- to 24-year-olds for landscaping training. Similarly, the Baltimore Step-Up program established a 12-week training partnership with the Baltimore Convention Center to identify, prepare, and train residents for jobs at the Convention Center. Pre-employment training is provided, in addition to career development, classroom training, and on-the-job focusing on aspects of catering and food service. Finally, one of the three Section 3 sites, Los Angeles, contracts with resident management corporation businesses to provide services at developments, including security. Rio Hondo, a school that specializes in law enforcement, provided training. During our site visits, we also observed efforts to train residents in high-demand jobs that fell outside of the discrete HUD programs under review. For example, the housing authority in San Antonio operated a program that trains residents for specific jobs within the housing authority.

For the clerk/typist program, there were approximately 10 to 12 vacancies within the housing authority each year due to promotions. After completing the training, residents who could type 45 words per minute were placed in temporary positions until full-time positions became available. The maintenance/helper program had the capacity to train 30 participants each year. The program involved training in carpentry, electrical, plumbing, masonry, heating/air conditioning, renovation, and cabinet making. The housing authority also referred residents to Project Quest, an organization that provided training in health care, information systems, manufacturing, maintenance and repair, and service technology. Project Quest staff conducted extensive labor market analysis to determine which industries and occupations to target. Most programs are affiliated with local community colleges and are two years in duration.

## O. Basic skills training

## 1) Summary of Research

Basic skills training is considered to be "general" education and is not targeted to a specific job, industry, or trade. In may involve remedial education, English as a Second Language (ESL), Adult Basic Education (ABE), and General Education Development (GED) training. According to the U.S. Department of Health and Human Services, almost half of all welfare parents lack a high school diploma or GED. A large body of research demonstrates that individuals with the lowest basic skills are the least likely to become employed or work steadily. This suggests that providing welfare recipients with basic education will increase their ability to obtain and retain employment.

However, a 1996 review of research on basic education programs for welfare recipients found that fewer than half of the programs increased participants' employment or earnings. When earnings did increase, it was usually due to increases in the number of hours worked; not higher paying jobs. In addition, research suggests that most basic education programs do not greatly improve educational attainment. Few of the programs improve participants' scores on tests of basic skills.

Basic Skills Training	HUD Program
Integrated element of comprehensive	YAP
program	Youthbuild
One of multiple services available	FIC
	HOPE VI
	Jobs Plus

#### 2) Summary of Housing Authority Activities

Basic education is a common program element. Consistent with the literature, no HUD program focused exclusively on basic skills education. The most common courses were GED preparation and English as a second language (ESL).

In some cases, basic education was integrated into a larger HUD program. YAP and Youthbuild, for example, treated basic education as a means of opening up future possibilities (e.g. college, jobs with increasing responsibility) and not as an end in itself. For example, all four YAP sites provided a combination of basic education and work skills. The education component entailed GED preparation or enrollment in a high school program, community service opportunities, and life skills/leadership training. YAP participants generally spent one or two days per week in the

classroom. The five Youthbuild sites we observed generally alternated from week to week between classroom-based basic education and on-site construction training.

Other HUD programs—FIC, Jobs Plus, and HOPE VI—offered more self-contained basic education courses. In these cases, staff suggested that basic education was an important step toward self-sufficiency, one that will make a candidate more employable. Even in these instances, however, the basic education courses were one of several activities that residents could select, along with job placement services and, often, vocational training. All three FICs and both Jobs Plus sites we observed offered basic skills education on site. GED preparation classes were the most common basic education courses; ESL was also popular in Los Angeles. All five HOPE VI sites also offered access to basic skills education. In Baltimore and Milwaukee, GED preparation, adult basic education courses, and computer training were available on site. Boston and Los Angeles referred residents to other resources in the community.

Other HUD programs, including FSS, Neighborhood Networks, EDSS, MTO, and TOP referred participants to basic skills education providers as needed.

## P. Post Secondary Education

## 1) Summary of Research

Post secondary education generally involves college-level courses after receiving a high school diploma of GED certificate.

Research suggests that attaining a post-secondary degree has a positive effect on an individual's ability to advance to a better-paying job. Those with a college education earn considerably more than individuals who have not attended college. Analysis of data from the National Longitudinal Survey of Youth indicate that hourly earnings for all women increase by 19 percent to 23 percent when they obtain an associate's degree, and by 28 percent to 33 percent when they obtain a bachelor's degree.

Under TANF, post-secondary education as a pre-employment service is less feasible. Postsecondary education directly related to employment cannot count towards the first 20 hours of participation unless the participant is a teen parent. However, post-secondary education can be combined with work as a strategy for wage progression and career advancement. Nonvocational post-secondary education generally does not count towards the participation requirements.

It is difficult to determine the effect of post-secondary education on the employment and earnings of welfare recipients because few employment and training programs offer it, and fewer have been evaluated. When post-secondary education is available, it is generally one of many components that could affect work effort and earnings.

Post-Secondary Education	HUD Program
Referrals	FSS
	Jobs Plus
Scholarships	YAP
	Youthbuild

## 2) Summary of Housing Authority Activities

Post-secondary education was one of the least common program elements we observed. One factor in the low level of post-secondary education was the nature of the welfare programs in the five sites. In only one site, Baltimore, did the welfare rules enable recipients to pursue college education.

Two programs referred participants to post-secondary education: FSS and Jobs Plus. In four of the five sites we observed, a number of FSS participants were involved in post-secondary education. In Baltimore, for example, 11 percent of FSS participants were enrolled in college. In San Antonio, the proportion was 20 percent. The program's escrow component encouraged college enrollment, because participants can use their escrow funds prior to completing the program to pay for post-secondary education.<sup>26</sup> One of the two Jobs Plus sites, Baltimore, also referred clients to colleges for short-term, customized training. The program staff had agreements with local community colleges to provide training courses as needed. The Maryland Center for Arts and Technology course for customer services and finance described earlier is an example of customized training.

Two programs, YAP and Youthbuild, offered Americorps scholarships for post-secondary education to program graduates. In Baltimore and Los Angeles, for example, about 25 percent of Youthbuild graduates used Americorps scholarships to enroll in college.

## Q. Assessment

## 1) Summary of Research

Assessment generally entails an interview with a case worker or other staff person to determine the general characteristics of the participant. Higher intensity assessment might include barrier screening or skills testing.

Assessment can be a useful tool to determine if recipients face major barriers to employment, to target specific services, and to gauge employment skills and interests. In theory, a good assessment would enable case workers to determine, at intake, the particular blend of services each participant will need.

Assessment also can help programs focus on those most in need of assistance. To the extent that those who are likely to leave welfare or maintain employment on their own can be identified, HUD staff can conserve their resources by minimizing services to this group and concentrating on those who are less likely to leave welfare or sustain long-term employment without intervention.

While assessment makes sense intuitively, there is little evaluation research to suggest that it can successfully identify who will benefit most from a particular service strategy.

Assessment	HUD Program
In-depth	FSS
	Jobs Plus
	МТО
Eligibility determination	YAP
	Youthbuild
Site-specific use	EDSS

## 2) Summary of Housing Authority Activities

<sup>&</sup>lt;sup>26</sup> Funds can also be used for other activities consistent with fulfilling the contract of participation.

	FIC
	Neighborhood Networks
	TOP
	DEG
Service design	HOPE VI
	Neighborhood Networks

Assessment was one of the most common program components observed. It occurs, in some form, in every HUD program except Section 3.

In programs with case managers, such as FSS, Jobs Plus, and MTO, assessment generally was in-depth, and gathered information on a client's educational attainment, work history, and interests. In FSS and Jobs Plus, assessment was used to craft a service plan or contract of participation, which detailed the services the client would need to become self-sufficient (e.g., job search, education, training, paid work experience, child care).<sup>27</sup> The MTO program used assessment to help residents prepare for their moves to low-poverty neighborhoods. Generally, counselors assessed the client's barriers to employment and the services that would be needed after the move.

In other programs, principally those that targeted youth, assessment was more general and involved screening for program eligibility, skill deficiencies, and substance abuse. For other HUD programs, the use of assessment varied by site. The EDSS programs in Boston and Milwaukee took different approaches to assessment. Boston plans to bring staff from the city's one-stop career centers on-site at selected developments to conduct basic vocational assessments. The Milwaukee housing authority, on the other hand, used the EDSS grant to fund resident employment coordinators who assessed the job readiness of adults through in-person interviews. Assessments focused on a client's educational attainment, previous work experience, job interests, welfare status, and access to Food Stamps, child care, and Medicaid. Two of the three FICs we observed also offered some form of assessment. In Los Angeles, the assessment reviewed a new client's educational background, work history, skill level, family needs, and interests, and was used to develop an Individualized Service Plan. In Baltimore, the nature of assessment depended upon the service the resident chose. Job seekers were assessed by the city's on-site Department of Labor (PACE program); other activities did not involve assessment. Similarly, one of the three Neighborhood Networks sites, Boston, assessed the math and reading skills of clients and collected background information, including educational attainment, work history, and barriers to employment, to determine whether a client needed basic skills education or supportive services, and where to begin the computer training program. In addition, one of three DEG sites, San Antonio, used skills assessment and interest surveys to determine whether participants in the Economic Development program needed specific training. Finally, one of three TOP sites, Milwaukee, assessed all new residents of the development targeted by the grant for physical and mental health problems, family situations, and financial circumstances.

<sup>&</sup>lt;sup>27</sup> In Milwaukee, the housing authority moved away from this model. Initially a community-based organization (YWCA) was under contract to provide case management and assessment. This function was cancelled due to low client participation in the FSS program because clients on welfare were required to meet with welfare agency case managers. Instead, housing authority rent assistance specialists provide clients with information regarding welfare, Medicaid, child care, or transportation to FSS as well as other (non-FSS) Section 8 tenants.

Finally, two HUD programs assessed resident needs to determine the mix and breadth of components. Two HOPE VI sites, Boston and Baltimore, used assessment to determine the service needs of residents and to select service providers. Similarly, one of the three Neighborhood Networks programs assessed potential participants<sup>28</sup> and found that the majority of respondents had never used a computer or the Internet. Based on this assessment, program staff determined that computer "literacy" (e.g., mouse use) needed to be a component of the training, and a two-week element was added to the course.

We also noted use of assessment outside of the realm of HUD programs under review. The housing authority in Los Angeles used assessment extensively in the Welfare-to-Work program. The assessment tool used for those with the lowest skills or education level,<sup>29</sup> CALIPER, was designed to reveal the area(s) of work a client was best suited for and then matched his or her skills to available jobs.

## R. Case Management

## 1) Summary of Research

Entering the workforce for the first time, or after a long absence, can be a daunting task. The services many welfare families need are numerous and provided by a vast network of agencies. Early intervention to address barriers and link participants to services is critical to getting welfare recipients employed, keeping them employed, and helping them find new jobs, particularly in light of the time limits and participation requirements of TANF.

Many programs use a case management approach to help welfare participants obtain the services they need. Case management can begin prior to employment and continue for some period after employment to reduce the frequency of job loss and, if necessary, facilitate re-employment. An evaluation of **Florida's Family Transition Program** compared two counties in which the eligibility and case worker roles were combined into one case manager position to counties where the roles were separate. The evaluation found that the clients had generally positive experiences with the combined case manager. Participants in Escambia County were more likely than controls to have talked with their case managers about specific topics, including mental health/counseling (51 percent vs. 4 percent) and housing (44 percent vs. 12 percent). The findings in Alachua County were similar.

A key component of the **Postemployment Services Demonstration** (PESD) projects was the provision of intensive case management to promote job retention and advancement among newly employed AFDC participants. According to the evaluator, almost 9 out of 10 PESD clients used at least one case management service during the first six months of the program. More than 70 percent received counseling and support services (including discussions of how to budget money and workplace behavior). Four in ten received job search assistance. The impact results, however, were quite disappointing; only one of the four sites had modest impacts on job retention, employment and earnings.

## 2) Summary of Housing Authority Activities

Case Management	HUD Program
Component in all sites	FSS

<sup>28</sup> Residents of Orchard Mews (a Section 8 development) and McCulloh Homes (a public housing development).

<sup>&</sup>lt;sup>29</sup> Those with a fifth grade reading level or below.

	Jobs Plus MTO
Component in some sites	EDSS
-	FIC
	HOPE VI
	Neighborhood Networks
	TOP
	YAP
	Youthbuild

All five housing authorities we reviewed used case management in some programs.

Case management was an established component of three HUD programs: FSS, Jobs Plus, and MTO. For example, it was a core component of the FSS program in four of the five sites. As noted in the previous section, an FSS case manager conducted an in-depth assessment of a participant's educational attainment and career goals to develop an individualized service plan. Case managers helped clients access services by providing referrals. They also met with clients periodically to determine their progress toward self-sufficiency. Depending on the site and needs of the client, these reviews were conducted by mail, telephone or in person; the number of contacts was based on the client's needs. Case management was also a component of both Jobs Plus sites we observed. The process generally began with the resident completing an intake form, which collected basic information about the household. The participant then met with a case manager, who conducted a more thorough assessment of the client's training and employment goals. Referrals were made for appropriate services. Case managers followed-up with clients once per week for the first few months after employment began, then bi-monthly. Finally, case managers, or mobility counselors, at both MTO sites provided an array of services, such as helping the client find new accommodations by locating properties, driving clients to interviews with landlords, and assisting with rental applications. Following the move, the counselor conducted an assessment of skills and education; and made referrals if services were needed or desired.

Other HUD programs incorporated case management on a site-by-site basis. For example, one of the two EDSS programs, Milwaukee, offered case management services. The resident employment coordinators met weekly with employment seekers to assist with job search activities and make referrals for supportive services (e.g., child care, transportation). After a resident was placed in a job, retention services and continued contact depended upon the needs of the resident. In addition, one of the three FICs, Los Angeles, had a case management component. As noted earlier, the case manager reviewed the educational background, work history, skill level, family needs, and interests of new clients. Referrals were made accordingly for training and supportive service needs.

Four of the five HOPE VI sites we observed had a case management component, although the intensity of services varied considerably. Three sites referred residents to community resources on an as-needed basis. Only the Baltimore program provided case management services for each resident in the HOPE VI development. The housing authority's Family Support Services division assigned every family to one of three on-site case managers. Each manager had an active caseload of approximately 35 cases. The intensity of services depended on the resident's needs. About half of the families met weekly with case managers.

Two of the three Neighborhood Networks sites offered informal case management. In Baltimore and Boston, staff members referred participants to other providers for supportive services. The Milwaukee TOP program also provided case management. The case managers assessed the needs of the elderly, substance abusers, residents with mental health problems, and employment seekers, and helped them obtain services from the community. The level of contact between case managers and residents varied depending on individual needs. At a minimum, case managers met monthly with residents and conducted six-month reviews of their initial assessments. Some form of case management was provided by four of the five Youthbuild sites. The level of intensity varied from follow-up of graduates to one-on-one planning. At a minimum, staff referred participants to services as needed. Case management was most intensive in Los Angeles, where a case manager on staff provided counseling and referrals to assist trainees with health care, housing, child care, and family emergencies that might otherwise interfere with their ability to succeed. Finally, some of the YAP sites noted that case management is offered informally to those who need it.

We also observed instances where case management primarily occurred outside of the HUD programs under study. The Los Angeles housing authority's Welfare-to-Work program had 10 case managers who typically met with or contacted their clients one to three times per week. Together, the case manager and client developed an Individual Service Strategy, defining both short and long term goals and identifying barriers to employment. Case managers also helped clients obtain other needed services, including child care, transportation assistance, drug and alcohol abuse treatment, and domestic violence counseling. Case managers continued to work with clients after job placement to help them retain employment and secure any other services that may be needed. In addition, the one-stop satellites had case managers on site to review the client's occupational skills, family history, education, work history, supportive service needs, goals, and general attitudes and motivation. Based on this assessment, the case manager drew up a service strategy.

### S. Barriers to Employment

#### 1) Summary of Research

Research suggests that personal and environmental barriers often limit employment prospects for welfare recipients. Personal barriers include domestic violence, mental health, and substance abuse. Examples of environmental barriers include child care and transportation. The Women's Employment Study (WES) found that women on welfare are more likely to experience a barrier than those not on welfare, and that the probability of being employed 20 or more hours per week declines as barriers increase. Starting in 1997, WES tracked a random sample of single mothers on welfare in an urban Michigan county to determine the prevalence of barriers to employment. The study examined lack of car/drivers' license, depression, alcohol dependency, drug dependency, mother's health problem, child's health problem, and domestic violence. In many cases, women on welfare were considerably more likely to have a barrier when compared with a national sample of women. For example, 27 percent of welfare recipients experienced depression, compared to 13 percent of the national sample. Those with depression were about 25 percent less likely to work 20 or more hours per week than those without the barrier (46 percent versus 61 percent).

When multiple barriers were present, the probability of employment declined further. For women in the sample with no barriers, the probability of working 20 or more hours a week was 78 percent; for women with one barrier, it was 70 percent. However, two or three barriers decreased the probability of working to 56 percent; four to six barriers decreased it to 36 percent.

Services Addressing Barriers to Employment	HUD Program
<b>On-site</b>	FIC
	HOPE VI
	Jobs Plus
Referrals	EDSS
-	FSS
	МТО
	Neighborhood Networks
	Step-Up
	TOP
	YAP
	Youthbuild
	DEG

#### 2) Summary of Housing Authority Activities

All HUD programs observed, with the exception of Section 3, offer barrier removal services, either directly or through referrals.

Three HUD programs offered on-site services: FIC, HOPE VI, and Jobs Plus. The most common services offered were child care and health services. For example, two of the three FICs we observed, Baltimore and Milwaukee, offered on-site child care services and health education. For example, one Baltimore FIC offered child care for children under age three if their parents are participating in center programs, before- and after-school child care (and vacation care), and a Head Start wrap around program. Similarly, the HOPE VI programs in Baltimore and Milwaukee offered child care (infant, toddler, and before/after school care) and full service health clinics on site. Finally, both Jobs Plus sites addressed possible barriers to employment, and strategies for overcoming them, during intake. Often, services were available on site. In Baltimore, for example, Visions for Health helped those without insurance to access health care. The Margaret Ruffin Center provided child care during non-standard hours. Also, the Jobs Plus staff was trained by the welfare agency to process child care vouchers on site, and bus tokens and passes were available.

A number of HUD programs offered services that address barriers to employment primarily through referrals. Both EDSS sites, for instance, referred clients to child care, transportation, remedial education, substance abuse prevention, and other supportive services as needed. In four of the five FSS programs we observed, the client's service plan or contract of participation addressed barriers to employment and the services that would be needed to overcome them. Other programs that offered referrals, primarily for child care and transportation, included MTO and Neighborhood Networks. The Step-Up program also provided child care to all participants, regardless of whether they were TANF clients. A Human Services Manager also was available to assist with housing problems, denial of medical assistance, and other emergencies. All YAP sites questioned participants about potential barriers to participation in the program during intake. Referrals were provided as needed for child care, transportation, low cost health services, assistance with legal dilemmas, and assistance finding affordable housing. Likewise, in all five Youthbuild sites, participants could get referrals to needed services, such as substance abuse or mental health on an as-needed basis.

Other programs focused on more personal barriers to employment, such as mental health issues and substance abuse. One of the three TOP programs, Milwaukee, focused on helping residents with mental illnesses or substance abuse problems to obtain services from community agencies. The grant funded an on-site nurse and social worker to provide general alcohol education to all residents and to facilitate access to treatment and compliance with treatment regimes. In addition, three sites use (or plan to use) the DEG to help eradicate substance abuse in developments and co-locate supportive services that will enable participants to work. The focus of the programs in two sites where the program is operating is somewhat different.<sup>30</sup> In San Antonio, staff notes that substance abuse flourishes in environments where people are not fully utilizing their time. Thus, the program seeks to offer alternatives for residents who might otherwise spend time involved with drugs, such as youth sports leagues and entrepreneurship programs for adults. In Baltimore, the two programs funded by the DEG aim to help people overcome substance abuse problems through referrals to health care services. Housing authorities also addressed barriers to employment outside the realm of HUD employment and training programs. The most common barriers addressed by the five housing authorities are child care and transportation.

In Milwaukee, the housing authority provides child care on site at all five family developments. It is available to all residents, not just welfare clients. The Baltimore housing authority is the largest provider of day care services in the city. There are 10 child development centers, which provide over 750 slots. Child care is provided on site at the FICs, and at the developments targeted by HOPE VI, Jobs Plus, and DEG. The housing authorities in Boston, Los Angeles, and San Antonio typically address the need for child care through referrals to providers, although a limited number of developments in Los Angeles have child care centers that are operated by resident management corporations.

Transportation was identified as a barrier by staff at four of the housing authorities. In Baltimore, Los Angeles, Milwaukee, and San Antonio, program staff helped clients secure transportation by offering tokens or bus passes. Transportation seemed especially problematic in Los Angeles, which does not have an extensive rail system to transport residents from areas of high unemployment to those with job opportunities. As a result, residents who do not own vehicles must rely primarily on buses, which can be both time consuming and complicated. Case managers of the housing authority's programs assisted their clients in obtaining transportation vouchers or tokens through the welfare department. The Boston housing authority staff was unique in its assessment that transportation is not a barrier to employment. Staff noted that most residents could reach employment or social service providers on the extensive public transportation system and the housing authority assisted with bus and subway passes and tokens. However, staff views psychological barriers as a larger deterrent to work than physical barriers. Many residents simply do not want to leave their developments or immediate neighborhoods.

## T. Cash and in-kind supports

# 1) Summary of Research

In the pre-TANF environment, policies to "make work pay," such as the Earned Income Tax Credit (EITC) were discussed in terms of making employment more attractive than welfare. Today, choosing between welfare and work is not an option. Policies such as the EITC serve a

<sup>&</sup>lt;sup>30</sup> The Boston program is still being developed.

different role: helping women to provide an adequate standard of living for their families and move toward self-sufficiency.

There are a number of policies that can help working families move out of poverty, such as the EITC and earned income disregards.

EITC. Numerous studies indicate that the Earned Income Tax Credit (EITC) encourages work. According to one study, the EITC lifted more children out of poverty than any other government program or category of programs. The Center for Budget and Policy Priorities (CBPP) determined that the number of single mothers who worked grew substantially between 1984 and 1996, and that half of this increase resulted from the EITC program and its expansion during this period. In 1996, according to CBPP, the EITC lifted 4.6 million people in low-income working families out of poverty. In 1997, over 19 million workers received credits totaling \$28 billion. Income Disregards. Disregarding (or not counting) a portion of earned income for the purpose of calculating welfare benefits can serve two purposes. It can encourage welfare recipients to work by allowing them to keep more of their earnings in addition to their welfare grants, thus increasing total income. It can also supplement the income of those who work in low-wage jobs. It is an attractive policy, since welfare recipients can gain work experience without losing their safety net, and possibly gain skills that might help secure a higher paying job in the future. Recent evaluations have shown promising results, although it is difficult to determine how much of an increase in work or earnings is due to the disregard versus other policy changes. For example, the Minnesota's Family Investment Program includes a 38 percent disregard of earnings as well as a 20 percent increase in the base TANF grant for workers. An evaluation of the program found that employment and earnings increased substantially, and that the positive impact for some groups was due largely to the program's financial incentives. Income disregards, however, can be a double-edged sword. In a time-limited welfare environment, a client who mixes work and welfare uses up a share of her lifetime benefits. Some states are exploring ways to exempt clients who mix work and welfare from time limits.

### 2) Summary of Housing Authority Activities

Cash and In-kind Supports	HUD Program
One component	FSS
	Jobs Plus
	МТО

Three HUD programs offered cash and/or in-kind supports to participants. FSS provided an escrow account; the funds were available to clients upon completion of their contract of participation. Housing authorities encouraged participants to use the funds for home ownership, post-secondary education, a car, or similar purposes that would aid in self-sufficiency, although the accounts can be used for any purpose. In Boston, the amount of savings FSS participants accrued in their escrow accounts ranged from \$4,600 to \$17,000.

A key component of the Jobs Plus program was enhanced financial incentives to encourage work. The two sites we observed took different approaches. Baltimore proposed decreasing the rent calculation for working families from 30 percent of earned income to 20 percent. The housing authority also plans to implement a rent ceiling (Riccio, 1999). Los Angeles is seeking approval from HUD for a rent incentive package that includes an 18-month rent freeze for working families, followed by flat rents. The housing authority is also considering a voluntary escrow account, to help facilitate the movement of Jobs Plus participants from welfare to work.

Finally, families that participated in MTO received a housing voucher that was redeemable only if they moved to a census tract where less than 10 percent of families were in poverty. In the two sites we observed, non-profit organizations also helped families obtain credit reports, fill out housing applications, and assemble monthly budgets.

In addition to the HUD programs under review, a principal cash support available to housing authorities was the disregarding of residents' earned income for the purposes of rent calculation. Just as welfare recipients who fear that increased earnings will negatively affect their welfare benefits might be less likely to work, residents who fear that earnings resulting from employment will result in a large increase in rent might be discouraged from working. Disregarding earnings when calculating rent could ease concerns. Federal law, in fact, required PHAs to disregard, for a set period, any increase in public housing residents' earnings resulting from participation in a public training and employment program. The Quality Housing and Work Responsibility Act (QHWRA) of 1998 requires housing authorities to disregard all increased income from employment for an initial 12-months and 50 percent of increased income for a subsequent 12-month period.

Conversations with housing authority staff revealed that the earnings disregard was not applied consistently. Only Baltimore appeared to have a policy similar to QHWRA. During the first year of a new job, the rent is not adjusted to reflect the increase in earned income. The freeze ends after one year. During the second year, rent cannot increase more than 50 percent of the allowable amount of the increase. The full income-adjusted rent increase goes into effect at the beginning of the third year.

In one of the five sites we visited, some portion of resident income is disregarded; in another, individual developments were taking the initiative to educate tenants about the HUD law. In Milwaukee, for example, residents received a \$2,000 income disregard when they found employment, and their rent did not increase until their following annual review. In Boston, staff at one development was working with Greater Boston Legal Services (GBLS) to educate residents about HUD's 18-month rent freeze policy for residents whose income increases as a result of job training. In 1999, notices were placed in rent statements alerting tenants that if they lived in public housing since October 1994 and received employment training and subsequently went to work, they were entitled to a rent freeze for 18 months after beginning their jobs. Other developments may follow suit.

Housing authorities also attempted to educate residents about the EITC. The Milwaukee housing authority, for example, disseminated information regarding tax credits in rent statements, as well as through the management office and agencies that are on-site at developments. In Baltimore, individual Resident Associations helped educate residents about cash and in-kind benefits by organizing meetings to discuss welfare reform, the EITC, and other supports. In Boston, one development placed information about EITC and other supports in the community center. Housing authorities made other in-kind supports available on a development by development basis. Two of the five family developments in Milwaukee, for example, had CBOs on site that operated food pantries where residents can get food free of charge or at reduced rates. In Boston, staff from one development worked with a local food bank to distribute food baskets to residents. The food baskets are available in the community center, and as a condition for receiving one, non-elderly residents had to peruse the job listings or the training program binder.

# **U. Transitional Supports**

# 1) Summary of Research

Welfare recipients often need to learn a variety of "soft skills" to help them cope with the work world. Participants face numerous challenges when they beginning work, including adapting to the new costs and demands of working; meeting the demands of the workplace (fitting into the workplace culture, meeting performance standards, and adjusting emotionally); dealing with the lack of support by family and friends; and finding a new job following job loss.

Some programs use life skills training and support groups to help clients transition from welfare to work.

**Life skills training**. Because many welfare recipients have little recent work history, they might be unfamiliar with the demands and requirements of the workforce. They might need assistance in learning how to balance work and family. Life skills training might include assistance with: budgeting, job etiquette, developing goals and recognizing personal challenges, changing destructive habits, building self-esteem, and peer support groups. Life skills training classes also help to identify those in need of more intensive services because program staff works closely with participants on a highly personal level. As a result, screening and other assessment tools previously discussed often take place during life skills sessions.

**Support groups**. Having someone, or a group, to talk to on an ongoing basis during the transition from welfare to work might make the process easier for some women. Although support groups and mentoring may be low-cost, common sense, supportive arrangements, little is known about the effectiveness of these interventions. One example is the West Humboldt Employment Training Center, located in a low-income Chicago neighborhood, which provides social support through support groups. Participants select from one of several support groups, depending on their personal and family needs. At weekly group meetings, participants share steps they have taken to meet explicit goals. According to staff, since the groups began operating, case managers have had to deal with fewer emergencies. They attribute the decline to participants' knowledge that they have a set time each week to discuss problems.

Transitional Supports	HUD Program
Central element	МТО
One program element	FIC
	FSS
	HOPE VI
	Jobs Plus
	Neighborhood Networks
	YAP
	Youthbuild
	DEG

## 2) Summary of Housing Authority Activities

All HUD programs, with the exception of Section 3, provided access to transitional supports. These supports were a central element of MTO. In both sites, participants received general transitional supports prior to moving to a low-poverty area. The life skills topics included assistance with credit reports, how to assemble a monthly budget, how to interact with landlords (what to wear, what to bring), and how to prepare for the move. After the move, housing counselors provide referrals to transitional services on an as-needed basis. Transitional supports were also a common aspect of seven other HUD programs. For example, in YAP and Youthbuild, transitional supports were often built into the larger education component. Examples of topics included appropriate communication skills, dealing with difficult supervisors, getting input from others, and conflict resolution. Programs that offered a constellation of services on site, such as Jobs Plus, HOPE VI, and FIC, often included transitional supports. All three FICs and the Baltimore HOPE VI site, for example, offered parenting education. Peer support groups were available at the Milwaukee FIC and the Baltimore Jobs Plus site. All FSS sites referred clients to transitional supports as needed. Some Neighborhood Networks and DEG sites offered transitional supports to clients with minimal or no work experience. The Boston Neighborhood Networks program, for example, had a life skills component that included training on office-related skills, such as how to talk on the telephone, to more personal problems such as dealing with credit, managing time, and budgeting. The San Antonio DEG program included a pre-employment component that focused on interpersonal skills, self-esteem, conflict resolution, and life skills training.

## CHAPTER V: SUMMARY OF FINDINGS AND CONCLUSIONS

### V. Summary of Findings

# 1) HUD programs are generally consistent with the literature on employment and training programs

HUD programs under review were generally consistent with promising welfare-to-work approaches identified in the literature and emphasized approaches and components adapted to a work-first environment. Most programs that offer basic skills education, training, or postsecondary education are related to the needs of the local labor market and are employment focused. Almost all programs stress job search or contain a job search component. However, two of the more effective program components, subsidized employment and on-the-job training, were not well represented in HUD programs implemented at the selected sites. Assessment and case management were among the most common program elements observed; however, while intuitively sensible, their connection to employment and self-sufficiency is not well established in published studies. Finally, almost all programs attempt to help residents overcome barriers to employment, such as child care, either directly through the provision of on-site services or through referrals.

#### a) Most HUD programs are employment focused

Three programs are employment focused: Section 3, Jobs Plus, and EDSS. While none excludes the option of training, the overarching focus of each program is work. Three training programs could be classified as education- and training-focused with a strong work element: Step-Up, YAP, and Youthbuild. They provide hands-on experience in a well-paying trade and often help place participants in jobs. Only two programs, FSS and Neighborhood Networks, could be classified as education- and training-focused initiatives.

The remaining HUD programs are not easily classified. These include FIC and HOPE VI, both of which provide space to housing authority and non-housing authority service providers. Services offered by these programs range from basic education to job search.

#### b) Job search is a key program component

Almost all programs offer some form of job search. For some, such as EDSS and Jobs Plus, job search is a central focus, and participants have access to both job-related and supportive services. In other programs, job search is one, albeit smaller, component of a larger intervention. Step-Up, YAP, and Youthbuild, for example, focus on training in specific fields, although participants often receive assistance finding jobs. Programs such as FIC and HOPE VI aim to provide a constellation of services to residents, and job search is one such activity. Other programs, such as FSS and MTO, refer clients interested in employment to other service providers. Neighborhood Networks also had job search activities in some sites.

### c) Training programs are generally linked to work

HUD vocational training programs range from long-term (one or more years) efforts that seek to prepare participants for specific careers or trades to short-term or customized training. Most of these HUD programs train residents for jobs in demand. Youthbuild, YAP, and Step-Up train participants for construction-related jobs. Because housing authorities spend millions of dollars each year on construction and renovation, there is a demand for skilled labor. Only in YAP was there a guaranteed job; the others sought to place participants in permanent jobs, however. Other HUD programs, such as FIC, HOPE VI and Jobs Plus, incorporate short-term or customized training programs that focus on high-demand jobs such as child care and health services on a site-by-site basis. Neighborhood Networks focuses on computer skills.

### d) Education is often one part of a larger intervention

No single HUD program focuses exclusively on basic skills education. Youthbuild, YAP, and Step-Up, for example, combine hands-on construction experience with basic education. Consistent with the literature, the educational component is viewed as a means of opening up future possibilities (e.g., jobs with increasing responsibility) and not as an end in itself. Other HUD programs, such as FIC, Jobs Plus, and HOPE VI, offer more self-contained basic education courses. In these cases, staff suggests that basic education is an important step toward self-sufficiency, one that will make a candidate more employable. Even in these instances, however, the basic education courses are one of several employment-focused activities that residents can select. FIC, Jobs Plus, and HOPE VI also offer job placement services and linkages to vocational training.

### e) Post-secondary education is one of the least common program elements

Staff from most of the HUD programs noted that referrals to post-secondary institutions are possible under program guidelines, but that they are infrequent. FSS is the HUD program most likely to refer participants to post-secondary education. The Jobs Plus program in Baltimore also had agreements with local community colleges to offer customized training as needed. One factor for the infrequent post-secondary education referrals is the structure of the welfare programs in the five sites. Only the Maryland TANF program counted post-secondary education as a TANF work activity. In other sites, a TANF beneficiary would have to combine education with work.

### f) Subsidized employment and on-the-job training are less common

These are relatively rare program elements, even though evaluations of subsidized employment and on-the-job-training programs found significant, positive impacts on employment and earnings. Three programs used these components. YAP and Section 3 used on-the-job training extensively. Subsidized employment, or using TANF grants to fund program services, was part of the Step-Up program. There appears to be an opportunity in some sites for housing authorities and welfare agencies to collaborate and use TANF funds to support HUD programs. In three of the five sites we observed subsidized employment and on-the-job training were allowable work activities under TANF rules. In Baltimore, for example, subsidized employment was used to fund a number of resident initiatives, in addition to Step-Up.

### g) Assessment can reveal barriers to employment

Almost all HUD programs have an assessment component. In some cases, assessment is used to screen for program eligibility and possible barriers to participation, including substance abuse, skills deficits, and learning disabilities. In other programs, assessment involves gathering information on a client's educational attainment, work history, goals, and barriers to employment, with the aim of creating a service plan. The plan may focus on steps a resident could take to secure employment, or it may focus on overcoming barriers by referring the participant to GED classes, ESL, or other service providers.

#### h) Case management is common, yet the link to employment is less clear

Case management is a central component of three HUD programs: FSS, Jobs Plus, and MTO. Other HUD programs incorporated case management on a site-by-site basis. For example, we observed case management services at some, but not all, HOPE VI, EDSS, Neighborhood Networks, Youthbuild, YAP, TOP, and FIC programs. Most often case management links a resident to supportive and/or educational services. To the extent that such services make a resident more employable, they are a positive step. However, they often fall short of a work-first orientation. Jobs Plus and EDSS had the strongest focus on employment.

*i)* Direct provision of barrier amelioration services varies by site

Assessment and case management are the primary methods for determining barriers to employment and helping residents access services. While all housing authorities visited provided referrals for services, a number of housing authorities provided on-site services or space for barrier amelioration providers, including child care and health services. Most often, on-site services were provided through the HOPE VI, Jobs Plus, and FIC programs. DEG is also used by housing authorities to co-locate supportive services at targeted housing developments. In addition, housing authorities offer barrier removal services outside of discrete HUD employment and training programs. In Milwaukee, for example, child care is available at all five family developments, and multiple developments have health care providers on site. In Baltimore, the housing authority is the largest provider of child care services in the city.

### 2) Fewer HUD employment and training programs serve Section 8 recipients

Families holding Section 8 vouchers and certificates have less access to employment-focused HUD programs since the three employment and training programs that target Section 8 recipients are generally not work-focused. FSS has historically promoted education and training as a means to self-sufficiency and has encouraged participants to obtain advanced degrees that will improve their long-term employability. MTO primarily focuses on providing opportunities for families to live in areas with low poverty, in the belief that these areas will offer greater job opportunities and generally improve the life chances of participants. MTO programs help families find and lease approved privately owned housing units. Post-move supports, including counseling, job development workshops, and referrals for social services and/or education and training are available but participation is not mandatory. With the implementation of local programs using the new welfare to work vouchers, there will be a much stronger focus on linking this tenant-based rental assistance with job search and placement programs.

An important issue to consider for programs that target these Section 8 recipients is the greater difficulty they may find in reaching these clients. Unlike residents in public housing developments, Section 8 recipients are often scattered throughout many neighborhoods. Welfare and labor department officials may find working with public housing developments a more attractive option because they provide access to a large number of TANF clients in one place. Individual private owners or managers of properties in which some residents receive tenant based Section 8 assistance may not have the capacity or inclination to establish linkages with welfare and labor agencies for those receiving TANF assistance. In these cases, access to PHA programs may need to be expanded to serve these residents.

For families receiving Section 8 project-based assistance in privately-owned, HUD insured multifamily developments, HUD's Neighborhood Networks Initiative offers computer-based training and educational programs. However, the range of services offered is highly variable, and the number of developments that have Neighborhood Networks centers is very limited.

## 3) TANF resident participation in HUD programs is uneven

Only one HUD program, Jobs Plus, primarily served residents on welfare in all sites observed. This is not surprising, since Jobs Plus was designed to help resident transition from welfare to work. Other programs, such as EDSS, FIC, FSS, HOPE VI, and YAP served TANF clients, though to a lesser extent.

One factor that hinders program participation for TANF clients is the welfare rules in the five sites. The welfare department in only one site, Baltimore, accepted participation in HUD programs as work activities. The welfare department in a second site, Los Angeles, allowed TANF recipients to participate in the housing authority's programs *only* if they were unable to find work during a mandatory three-week job club. In the remaining three sites, Boston, Milwaukee, and San Antonio, welfare departments operated work-first programs that focused on employment at the earliest possible time. Consequently, acceptable work activities were those focused primarily on job search, and participation in education and training programs, including those offered by housing authorities, did not count.

Another factor that appears to limit participation in many HUD programs is the nature of the training. Step-Up, YAP, and Youthbuild are heavily focused on construction-related occupations. According to the staff at a number of housing authorities, many single mothers are not attracted to manual labor, while others find it difficult to arrange child care to deal with the long hours and travel requirements of construction work.

# 4) Programs that target TANF recipients combine HUD funding streams and harness resources from the broader community

A number of the local efforts observed combined HUD program funding streams and community resources to create employment and training programs that are locally designed and targeted toward residents on welfare.

#### a) Jobs Plus pools resources from other agencies

The Jobs Plus required housing authorities to collaborate with welfare and labor agencies to fund and design a program to help TANF recipients transition from welfare to work. In Baltimore, for example, the \$200,000 Jobs Plus grant (awarded to each of the seven national Jobs Plus sites) was matched by \$250,000 from the housing authority, \$80,000 from the local labor department, \$100,000 from the Empowerment Zone, and \$50,000 from the local welfare department. The welfare department worked with the housing authority to craft program activities that will count toward the TANF work requirement. Thus, TANF recipients have a range of services available. The labor department is on site to help place participants in jobs. The on-site child care center facilitates the transition to work. Residents who are not job-ready, or want to improve their job skills, work with on-site case managers to access activities such as on-the-job training, work experience, and education programs.

#### *b)* Housing authorities combined HUD funding streams to create employmentfocused programs tailored to community needs.

Housing authorities also adapted existing HUD programs to serve TANF clients and other employment seekers. This often involved combining HUD funding streams or tapping into labor and welfare department resources. In Milwaukee, for example, the housing authority used EDSS and HOPE VI funds to finance resident employment coordinators. These staff persons are the point people for residents seeking work. They conduct the assessments and serve as case managers and job developers. In two of the five developments targeted by the program, the resident employment coordinators also work with staff from the county welfare department and the "W-2 agencies" that administer the TANF program and the city's one-stop job centers. Coordinators work in tandem with W-2 staff on job search assistance; county staff assists with Food Stamp, Medicaid, and child care issues.

In San Antonio, the housing authority worked with the labor and welfare departments to establish one-stop resource centers at nine housing developments. The need for a service delivery system for the pending HOPE VI projects was the impetus for creating this model. While the housing authority's two HOPE VI developments are being constructed, residents of the public housing projects demolished to make way for the new projects have been temporarily relocated. The housing authority wanted to provide services to these residents in the interim, and developed a one-stop approach that became a model for a larger initiative. The housing authority partnered with the welfare and labor departments to create the Resident Opportunities and Achievement and Development (ROAD) Centers. It is providing space at nine developments for the Centers. In addition, FSS coordinators will be housed at the Centers to assist with counseling and supportive service referrals. The on-site welfare agency staff will provide eligibility determinations for TANF, Food Stamps, and Medicaid. The labor department will provide the employment services, including job readiness training, job placement, and post-employment services.

# c) The Los Angeles housing authority's primary programs for TANF clients is outside of HUD programs under review

In Los Angeles, the primary employment programs for residents on TANF fall outside of the discrete HUD programs under review. The Welfare to Work program is funded by a competitive welfare-to-work grant from the U.S. Department of Labor. The local labor department provides space at its one-stop career centers for the housing authority's welfare-to-work case managers, and the Private Industry Council, which funds employment and training services in the community, provided funding to serve additional residents. The Welfare-to-Work program is designed to operate in a work-first environment. Staff assesses clients to determine career skills, develop a service plan, and assist with job search. TANF clients can receive their work-related services through the program so long as the welfare agency certifies eligibility. The local department of labor also funds "satellites" to the one-stop career centers at a number of housing developments.

# 5) Strong linkages exist with labor agencies; relationships with welfare departments could be strengthened

We found varying levels of collaboration between housing authorities and the welfare and labor agencies in their communities. In all five sites, the housing authorities had a history of collaboration with the labor agencies. In Baltimore, for example, housing authority and labor department staff worked together to create the People Accessing Continued Employment (PACE) program at four developments to provide pre- and post-employment services to residents. In contrast, collaboration in Boston was more limited. In the past, the labor department simply funded outreach workers to inform residents about employment and training services in the community.

However, linkages with the welfare agencies appeared to be underdeveloped in many sites. While Baltimore and Milwaukee housing authorities have worked with welfare agencies in the past to design and administer programs to serve public and assisted housing residents, relationships and joint planning efforts have just begun in Los Angeles and San Antonio. In Boston, there is no coordination of employment services among high level housing authority and welfare department staff; although some coordination occurs between individual developments and local offices.

### 6) The implementation of financial incentives for residents must be resolved

HUD employs financial incentives to encourage public housing residents to work or seek training. The key HUD policy is contained in the 1998 Quality Housing and Work Responsibility Act (QHWRA), which replaced the 18-month mandatory income disregard with a 24-month disregard for residents whose income increases as a result of work or training. Under the new policy, housing authorities are required to disregard all increases in income from employment for an initial 12 months and 50 percent of increased income for a subsequent 12-month period. In addition to QHWRA, some individual housing authorities have similar financial incentives in place to encourage employment. Milwaukee, for example, disregards the first \$2,000 of earned income when a resident becomes employed and does not increase the monthly rent until the following annual review.

The negotiated rulemaking process, which will clarify QHWRA policy, was incomplete at the time of our site visits. We found significant confusion among PHA staff as to the nature of the

earned income disregard spelled out under QHWRA and its impact on PHA revenue. At the Baltimore site, the housing authority had implemented a policy that matched the QHWRA guidelines, but in the other four sites, the housing authority staff we met with appeared to be aware of statutory change but unsure about how it will be implemented, in particular, whether and how operating subsidies would be adjusted to cover rent losses. In one case, the Los Angeles authority had asked HUD to review their income disregard policies and expressed concern about going too far without direction from HUD and clarification of how rent losses will be treated in the current year and in the future.

Nor did residents we interviewed appear to know the rules governing HUD's policies on income disregards, and we did not observe systematic efforts on the part of housing authorities to educate residents about the policy. In one site, Boston, a non-profit organization was working with the tenant organization in one development to educate residents about the pre-existing 18-month earnings disregard. This lack of clarity appears to be affecting PHA implementation of the policies, as well as resident benefits from participating in employment and training programs.

# 7) Housing authorities approach issue of helping residents attain self-sufficiency differently

The five housing authorities we observed also take varying approaches to helping residents attain self-sufficiency. On one end of the spectrum, the Baltimore and Los Angeles housing authorities take a comprehensive approach by providing a broad array of employment and training programs, as well as on-site support services such as child care. The Los Angeles housing authority buttresses this approach with a strong emphasis on using its own construction company, Kumbaya, and resident management corporation-owned businesses for renovation and modernization projects.

On the other end of the continuum, the Boston housing authority adopted a more limited and decentralized approach to service delivery, guided by three principles: residents should use services located off site; the housing authority's role is primarily as a referral agent to other service providers; and, individual tenant organizations should take the lead on addressing the needs of their residents.

Falling in the middle, the Milwaukee and San Antonio housing authorities operate in strong work-first welfare environments. Both take the position that housing providers should help residents access employment and training services. In both sites, residents are often referred to outside providers. The Milwaukee housing authority also recognized that bringing community-based organizations on-site built important bridges to the community. Each family development has a number of on-site providers offering work supports ranging from child care to health services and food assistance. San Antonio is moving in the direction of on-site services. In summer 1999, the first ROAD Center opened at a housing development, and eight more will follow.

# W. Conclusions

HUD has a long-standing interest in promoting the self-sufficiency of residents in public and assisted housing. Crafting new policies at the national level, however, is difficult, because housing authorities have a great deal of flexibility in structuring employment and training services for their residents. While this flexibility is important to meet local needs, it makes it

difficult to create standardized programs. Housing authorities determine whether services will be offered by their staff or others in the community; whether services will be offered on-site at developments or through referrals to community organizations; and even whether services will be provided at all. Housing authorities also decide whether and the degree to which they will work with local welfare and labor authorities.

Despite these variations, our observations and interviews with staff during the site visits suggest a number of steps HUD could take to strengthen employment and training services for residents. First, HUD could facilitate collaboration among housing authorities, welfare departments, and labor departments. HUD activities could include providing PHAs with technical assistance or training on collaborations, encouraging PHAs to focus on retention and advancement services for employed residents, including incentives for collaborations in HUD's assessment systems, and helping provide a stable funding platform for collaborations. Second, HUD could take steps to ensure that PHAs have implemented the financial incentives for working residents outlined in the 1998 Quality Housing and Work Responsibility Act. Finally, HUD could address the lack of participation in certain programs among residents on TANF by determining barriers to participation and adjusting programs accordingly. If this proves difficult, HUD could consider ways to target non-custodial parents for these programs.

# 1) PHA employment and training activities for TANF residents could be designed to facilitate and support resident participation in welfare and labor funded services

Housing authorities, and welfare and labor agencies, share a common goal of helping TANF recipients gain employment. A large proportion of non-elderly families in public and assisted housing receives TANF. Thus, housing authorities could lose rent revenue if residents lose their welfare benefits and fail to find or maintain employment. The welfare and labor agencies are responsible for helping TANF recipients access work and supportive services. All three agencies not only have a vested interest in helping residents find jobs but also in helping them stay employed. Clients who cannot maintain employment will quickly exhaust their lifetime TANF benefits.

Housing authority programs that target TANF recipients, such as Jobs Plus, the Welfare-to-Work program, the San Antonio one-stop Resident Opportunities for Achievement and Development (ROAD) Centers, and the resident employment coordinators, involve collaboration with welfare and labor departments. In some cases, these agencies provide funding; in others, staff. In all cases, the housing authorities leverage resources from these outside agencies. This enabled them to conserve resources, which is important given their limited funding for employment and training and supportive services. Welfare and labor agencies, on the other hand, have considerable resources for these activities. In addition to TANF block grant funds, there are a host of financial resources in the community, including Workforce Investment Act (WIA) funds, Welfare-to-Work funds, and other state and local funds. The housing authority could assure access to a broad array of services for residents if it leveraged the resources available in the community. The question for housing authorities, then, is how they can augment and facilitate resident access to services provided by the welfare and labor agencies.

Housing authorities make attractive partners for their welfare and labor counterparts for a number of reasons. They provide access to a large number of TANF recipients in single locations, and may often be the first point of contact for residents seeking information about changes in the welfare system. As such, they are a resource for information on welfare and a

source of referrals to welfare agencies and employment and training programs. Housing authorities can also offer space for on-site service providers, including welfare and labor staff. In addition, they can serve as "vendors" for welfare or labor department-funded employment and training services. Finally, housing authorities can provide opportunities at their developments for residents on TANF to meet their work requirements. For example, TANF recipients who cannot find unsubsidized employment could be offered community service jobs in the areas of maintenance, construction, or child care services. Housing authorities could also target available employment opportunities specifically to residents at risk of exhausting their benefits.

HUD can take meaningful steps to facilitate collaboration among housing authorities, welfare departments, and labor agencies.

# *a) HUD activities could support this strategy by providing appropriate written materials, training, and technical assistance to PHAs*

While housing authorities were selected for this study because each was operating several of HUD's employment and training initiatives under review, we also found variations in their level of knowledge of welfare policies, such as work requirements and acceptable activities, and in collaboration with welfare and labor agencies. To help housing authorities become more involved in helping residents meet work requirements, HUD could provide more information to PHAs on welfare reform, the roles of welfare and labor agencies, and the sources and uses of funds for employment and training services. This could include written materials on welfare policies at the state level, training on how to create partnerships with labor and welfare agencies, and technical assistance to help individual housing authorities develop strategies to fund employment and training activities. HUD could work with housing authorities that have established relationships with their welfare and labor counterparts to educate others about successful practices.

#### b) HUD could encourage PHAs to focus on retention and advancement services

TANF recipients who begin work for the first time often need help retaining their jobs and advancing in the labor market. Welfare departments are currently working to design initiatives in this area. One challenge is finding ways to encourage working clients to continue to access services and work supports. Welfare agencies often lose contact with clients after they begin working. Housing authorities are in an ideal position to address this service gap. Developments offer access to a large number of current and former TANF clients. Often space is available to co-locate staff from the welfare or labor departments. Using the Baltimore example, the People Accessing Continued Employment (PACE) program, designed by the housing authority and the city's labor department, provides pre-placement, post-placement, and job-replacement services on-site at a number of housing developments. In terms of advancement, efforts to help low-wage workers combine work and education could help them advance in the labor force. Housing developments also offer on-site child care.

HUD can give retention and advancement services visibility by encouraging housing authorities to meet with welfare agencies to discuss joint efforts. The housing authorities could receive funding from the TANF agencies to provide post-employment services. In addition, HUD can encourage housing authorities to participate in ongoing evaluations of retention and advancement services. The U.S. Department of Health and Human Service's Administration for Children and Families, for example, has made job retention and advancement a major priority. Housing

authorities with retention programs (e.g. PACE) could join with their respective welfare agencies to develop strategies in this area.

# c) HUD could include incentives for collaborative employment and training programs in the new PHAS and SEMAP assessment systems.

HUD has developed new assessment tools for public housing, the Public Housing Assessment System (PHAS), and for housing authority administration of tenant based Section 8 programs, the Section Eight Management Assessment Program (SEMAP). The PHAS assessment is a tool to determine the extent to which housing authorities will be deregulated and permitted wider discretion in management and operation of public housing. A high-performing authority may also be eligible for bonus points in certain grant competitions. The assessment system references self-sufficiency activities under criteria addressing resident satisfaction, but almost all of the rule language speaks to customer satisfaction surveys, not to employment and training. SEMAP attempts to implement a similar concept. By adding employment and training efforts as a specific factor in these assessment frameworks, HUD could increase the motivation of local agencies to expand their linkages to local welfare and labor agencies.

# *d) HUD and PHAs could consider establishing funding mechanisms to provide a stable platform for PHA collaboration with local welfare and labor agencies*

Effective collaboration among the housing authorities and local welfare and labor agencies is time consuming for all involved. Staff time for planning and developing both informal relationships and formal cooperative agreements is a cost to housing authorities that, with a few exceptions, is paid from already over-subscribed operating and capital funds. In most cases, existing Resident Services budgets must cover what are essentially added administrative costs brought on by welfare reform. Some authorities may be able to cover a portion of these costs with Drug Elimination Grant funds, to the extent the activities are related to DEG program goals. Similarly, these activities may be funded as part of the cost of HUD-funded Family Self-Sufficiency coordinators in smaller PHAs.

HUD's Jobs Plus demonstration program is an example of a program explicitly designed to bring about collaboration between housing authorities and the key players in the new welfare system. Most of the funding needed to implement the demonstration was provided from foundations, with limited HUD funding. Authorities that received demonstration grants are expected to work closely with welfare and labor agencies to bring those agencies and their funded services to demonstration developments. While this model is still being evaluated, the collaboration envisioned makes sense intuitively, given the potential economic consequences of new TANF rules for both residents and housing authorities. Other agencies also are adopting this strategy. For example, the Department of Labor's Welfare to Work grant program (from which the housing authority in Los Angeles was awarded a competitive grant) emphasizes collaboration in planning and coordinating activities among numerous agencies involved in the program. HUD could recognize the administrative cost to housing authorities of creating collaborative relationships and find appropriate funding mechanisms to support these activities. At a minimum, these costs could be more explicitly recognized in public housing operating cost projections. In the future, the determination of administrative fees for the Section 8 voucher programs could also take account of the administrative costs of welfare-to-work related activities that benefit Section 8 recipients.

From a policy perspective, these public housing administrative costs could be addressed in the ongoing negotiated rulemaking on the new operating fund called for in QHWRA and funding included in HUD's future budgets. Along with the proposed funding of income disregards, this would provide important resources to help PHAs meet the QHWRA goals of transforming public housing developments into communities of residents who work.

#### 2) HUD could move aggressively to publish regulations on financial incentives to work and provide the necessary training and technical assistance to help PHAs implement the policy

The Quality Housing and Work Responsibility Act (QHWRA) provides a number of tools to housing authorities to help them establish policies that will encourage unemployed residents to find work and working families to remain in public housing. These include policies that establish ceiling rents below current "economic rent" levels to attract working families, and give residents the choice of paying flat rents or income-based rents. The Act also outlines a new mandatory income disregard policy for residents whose income increases due to training or work. The new policy replaces HUD's previous 18-month income disregard policy with a two stage, 24-month disregard. PHAs must disregard all increased income from employment for an initial 12 months and 50 percent of increased income for a subsequent 12-month period. Our site visit interviews revealed that income disregards have been implemented unevenly, and that housing authority staff and residents alike are often confused about the rules and their impact on tenant rent payments.

# 3) Current HUD programs that do not appear to serve TANF recipients could be reconfigured to attract more single mothers or redirected to focus on non-custodial parents

TANF client participation in a number of HUD programs appears to be limited. These programs include YAP, Youthbuild, Step-Up, and Section 3. Housing authority staff described a variety of reasons for low participation, including the nature of construction work and the difficult hours required for single parents with children. HUD could help increase participation among single mothers by determining the barriers to participation and adapting the programs accordingly. If the long hours away from home often associated with construction are preventing participation, HUD could help housing authorities focus on jobs in the community that do not require a long commute, such as maintenance work. If single mothers find some aspects of construction work more appealing than others, programs could accommodate these interests.

If reconfiguring programs appears to be difficult or not possible, housing authorities could retarget existing programs toward non-custodial fathers. Site visits indicated that the majority of participants in the construction-related programs were men. Staff could not determine what proportion were non-custodial fathers. There was a general sense that many non-custodial fathers would be precluded from program participation because many are not on leases; others have criminal backgrounds. A policy of serving non-custodial fathers would also be consistent with the initiatives in the broader labor environment, including the Department of Labor's welfare-to-work grants and the Workforce Investment Act, that make serving this population a priority.

HUD could provide written materials and technical assistance on how to involve non-custodial fathers in its employment and training programs, addressing such issues such as past criminal activity and leases.

#### BIBLIOGRAPHY

Administration for Children and Families, U.S. Department of Health and Human Services (1999). *Change in Welfare Caseloads Since Enactment of New Welfare Law*. ACF website: <u>http://www.acf.dhhs.gov/news/stats/aug-sept.htm</u>. August.

Center for Budget and Policy Priorities (1998). Impact of Quality Housing and Welfare Reform Act of 1998 on Welfare Reform Efforts. Washington, D.C.: CBPP.

Fishman, M., Barnow, B., Gardiner, K., Murphy, B., & S. Laud (1999). *Job Retention and Advancement Among Welfare Recipients: Challenges and Opportunities*. Washington, D.C.: Administration for Children and Families, Department of Health and Human Services.

General Accounting Office (1998). *Welfare Reform: Effect on HUD Housing Subsidies Difficult to Estimate*. Washington, D.C.: Government Accounting Office. GAO/RCED-99-14. December.

Khadduri, J., Shroder, M. & B. Steffan (1998). *Welfare Reform and HUD-Assisted Housing: Measuring the Extent of Needs and Opportunities.* Presented at "Managing Affordable Housing Under Welfare Reform: Reconciling Competing Demands," conference sponsored by Fannie Mae Foundation and the Center for Budget and Policy Priorities. June.

Kingsley, G. & Tatia, P. (1998). "Federal Housing Assistance and Welfare Reform: Uncharted Territory," *New Federalism Issues and Options*. Washington, D.C.: Urban Institute, Policy Brief A-19.

Lewin Group (1998). Policy Evaluation of the Overall Effects of Welfare Reform on SSA *Programs: Literature Review and Study Design Report.* Report prepared for the Social Security Administration.

Newman, S. (ed.) (1999). *The Home Front: Implications of Welfare Reform for Housing Policy*. Washington, D.C.: Urban Institute Press.

Newman, S. & J. Harkness (1999). "The Effects of Welfare Reform on Housing: A National Analysis," in *The Home Front: Implications of Welfare Reform for Housing Policy*, Sandra Newman, ed. Washington, D.C.: Urban Institute Press.

Riccio, J. (1999). *Mobilizing Public Housing Communities for Work: Origins and Early Accomplishments of the Jobs-Plus Demonstration*. New York: Manpower Demonstration Research Corporation. September.

U.S. Department of Housing and Urban Development, Office of Policy Development and Research (1998). *HUD Research Fact Sheet*. Washington, D.C.: HUD.

U.S. Department of Housing and Urban Development, Office of Policy Development and Research (1998). *Welfare Reform Impacts on Public Housing: A Preliminary Forecast*. Washington, D.C.: HUD. March.

U.S. Department of Labor, Employment and Training Administration (1998). *Workforce Investment Act of 1998*.

U.S. House of Representatives, Committee on Ways and Means (1998). *Green Book*. Washington, D.C.: Government Printing Office.

U.S. Department of Housing and Urban Development Office of Policy Development and Research Washington, DC 20410-6000

**Official Business** 

Penalty for Private Use \$300

Return Service Requested

